# Antibiotics to prevent complications following tooth extractions (Review)

Lodi G, Figini L, Sardella A, Carrassi A, Del Fabbro M, Furness S



This is a reprint of a Cochrane review, prepared and maintained by The Cochrane Collaboration and published in *The Cochrane Library* 2012, Issue 11

http://www.thecochranelibrary.com

# WILEY

# TABLE OF CONTENTS

HEADER
ABSTRACT
PLAIN LANGUAGE SUMMARY
SUMMARY OF FINDINGS FOR THE MAIN COMPARISON       4
BACKGROUND
OBJECTIVES
METHODS
Figure 1
RESULTS
Figure 2
Figure 3
DISCUSSION
Figure 4
AUTHORS' CONCLUSIONS
ACKNOWLEDGEMENTS
REFERENCES
CHARACTERISTICS OF STUDIES
DATA AND ANALYSES
Analysis 1.1. Comparison 1 Antibiotic versus placebo, Outcome 1 Local sign of infection
Analysis 1.2. Comparison 1 Antibiotic versus placebo, Outcome 2 Dry socket.
Analysis 1.3. Comparison 1 Antibiotic versus placebo, Outcome 3 Pain (dichotomous on 6th-7th day).
Analysis 1.4. Comparison 1 Antibiotic versus placebo, Outcome 4 Pain score (VAS 7th day)
Analysis 1.5. Comparison 1 Antibiotic versus placebo, Outcome 5 Fever (6th-7th day)
Analysis 1.6. Comparison 1 Antibiotic versus placebo, Outcome 6 Swelling (7th day).
Analysis 1.7. Comparison 1 Antibiotic versus placebo, Outcome 7 Trismus (7th day)
Analysis 1.8. Comparison 1 Antibiotic versus placebo, Outcome 8 Adverse effects.       62
ADDITIONAL TABLES         62
APPENDICES
FEEDBACK
WHAT'S NEW         70
CONTRIBUTIONS OF AUTHORS
DECLARATIONS OF INTEREST
SOURCES OF SUPPORT
DIFFERENCES BETWEEN PROTOCOL AND REVIEW
INDEX TERMS         . <th< td=""></th<>

[Intervention Review]

# Antibiotics to prevent complications following tooth extractions

Giovanni Lodi<sup>1</sup>, Lara Figini<sup>2</sup>, Andrea Sardella<sup>1</sup>, Antonio Carrassi<sup>1</sup>, Massimo Del Fabbro<sup>3</sup>, Susan Furness<sup>4</sup>

<sup>1</sup>Dipartimento di Scienze Biomediche, Chirurgiche e Odontoiatriche, Università degli Studi di Milano, Milan, Italy. <sup>2</sup>Milan, Italy. <sup>3</sup>Dipartimento di Scienze Biomediche, Chirurgiche e Odontoiatriche, Università degli Studi di Milano, IRCCS Istituto Ortopedico Galeazzi, Milan, Italy. <sup>4</sup>Cochrane Oral Health Group, School of Dentistry, The University of Manchester, Manchester, UK

Contact address: Giovanni Lodi, Dipartimento di Scienze Biomediche, Chirurgiche e Odontoiatriche, Università degli Studi di Milano, Via Beldiletto 1/3, Milan, 20142, Italy. giovanni.lodi@unimi.it.

Editorial group: Cochrane Oral Health Group. Publication status and date: Edited (no change to conclusions), comment added to review, published in Issue 5, 2013. Review content assessed as up-to-date: 25 January 2012.

**Citation:** Lodi G, Figini L, Sardella A, Carrassi A, Del Fabbro M, Furness S. Antibiotics to prevent complications following tooth extractions. *Cochrane Database of Systematic Reviews* 2012, Issue 11. Art. No.: CD003811. DOI: 10.1002/14651858.CD003811.pub2.

Copyright © 2013 The Cochrane Collaboration. Published by John Wiley & Sons, Ltd.

### ABSTRACT

#### Background

The most frequent indications for tooth extractions are dental caries and periodontal infections, and these extractions are generally done by general dental practitioners. Antibiotics may be prescribed to patients undergoing extractions to prevent complications due to infection.

#### Objectives

To determine the effect of antibiotic prophylaxis on the development of infectious complications following tooth extractions.

#### Search methods

The following electronic databases were searched: the Cochrane Oral Health Group's Trials Register (to 25 January 2012), the Cochrane Central Register of Controlled Trials (CENTRAL) (*The Cochrane Library* 2012, Issue 1), MEDLINE via OVID (1948 to 25 January 2012), EMBASE via OVID (1980 to 25 January 2012) and LILACS via BIREME (1982 to 25 January 2012). There were no restrictions regarding language or date of publication.

#### Selection criteria

We included randomised double-blind placebo-controlled trials of antibiotic prophylaxis in patients undergoing tooth extraction(s) for any indication.

#### Data collection and analysis

Two review authors independently assessed risk of bias for the included studies and extracted data. We contacted trial authors for further details where these were unclear. For dichotomous outcomes we calculated risk ratios (RR) and 95% confidence intervals (CI) using random-effects models. For continuous outcomes we used mean differences (MD) with 95% CI using random-effects models. We examined potential sources of heterogeneity. The quality of the body of evidence has been assessed using the GRADE tool.

#### Main results

This review included 18 double-blind placebo-controlled trials with a total of 2456 participants. Five trials were assessed at unclear risk of bias, thirteen at high risk, and none at low risk of bias. Compared to placebo, antibiotics probably reduce the risk of infection in patients undergoing third molar extraction(s) by approximately 70% (RR 0.29 (95% CI 0.16 to 0.50) P < 0.0001, 1523 participants, moderate quality evidence) which means that 12 people (range 10-17) need to be treated with antibiotics to prevent one infection following extraction of impacted wisdom teeth. There is evidence that antibiotics may reduce the risk of dry socket by 38% (RR 0.62 (95% CI 0.41 to 0.95) P = 0.03, 1429 participants, moderate quality evidence) which means that 38 people (range 24-250) need to take antibiotics to prevent one case of dry socket following extraction of impacted wisdom teeth. There is also some evidence that patients who have prophylactic antibiotics may have less pain (MD -8.17 (95% CI -11.90 to -4.45) P < 0.0001, 372 participants, moderate quality evidence ) overall 7 days after the extraction compared to those receiving placebo, which may be a direct result of the lower risk of infection. There is no evidence of a difference between antibiotics and placebo in the outcomes of fever (RR 0.34, 95% CI 0.06 to 1.99), swelling (RR 0.92, 95% CI 0.65 to 1.30) or trismus (RR 0.84, 95% CI 0.42 to 1.71) 7 days after tooth extraction.

Antibiotics are associated with an increase in generally mild and transient adverse effects compared to placebo (RR 1.98 (95% CI 1.10 to 3.59) P = 0.02) which means that for every 21 people (range 8-200) who receive antibiotics, an adverse effect is likely.

#### Authors' conclusions

Although general dentists perform dental extractions because of severe dental caries or periodontal infection, there were no trials identified which evaluated the role of antibiotic prophylaxis in this group of patients in this setting. All of the trials included in this review included healthy patients undergoing extraction of impacted third molars, often performed by oral surgeons. There is evidence that prophylactic antibiotics reduce the risk of infection, dry socket and pain following third molar extraction and result in an increase in mild and transient adverse effects. It is unclear whether the evidence in this review is generalisable to those with concomitant illnesses or immunodeficiency, or those undergoing the extraction of teeth due to severe caries or periodontitis. However, patients at a higher risk of infection are more likely to benefit from prophylactic antibiotics, because infections in this group are likely to be more frequent, associated with complications and be more difficult to treat. Due to the increasing prevalence of bacteria which are resistant to treatment by currently available antibiotics, clinicians should consider carefully whether treating 12 healthy patients with antibiotics to prevent one infection is likely to do more harm than good.

#### PLAIN LANGUAGE SUMMARY

#### Antibiotics to prevent complications following tooth extractions

Tooth extraction is a surgical treatment to remove teeth that are affected by decay or gum disease (performed by general dentists). The other common reason for tooth extraction, performed by oral surgeons, is to remove wisdom teeth that are poorly aligned/developed (also known as impacted wisdom teeth) or those causing pain or inflammation.

The risk of infection after extracting wisdom teeth from healthy young people is about 10%; however, it may be up to 25% in patients who are already sick or have low immunity. Infectious complications include swelling, pain, pus drainage, fever, and also dry socket (this is where the tooth socket is not filled by a blood clot, and there is severe pain and bad odour). Treatment of these infections is generally simple and involves patients receiving antibiotics and drainage of infection from the wound.

This review looks at whether antibiotics, given to dental patients as part of their treatment, prevent infection after tooth extraction. There were 18 studies considered, with a total of 2456 participants who received either antibiotics (of different kinds and dosages) or placebo, immediately before and/or just after tooth extraction. There were concerns about aspects of the design and reporting of all the studies. In all of the studies healthy people had extractions of impacted wisdom teeth done by oral surgeons.

This review provides evidence that antibiotics administered just before and/or just after surgery reduce the risk of infection, pain and dry socket after wisdom teeth are removed by oral surgeons, but that using antibiotics also causes more (generally brief and minor) side effects for these patients. Additionally, there was no evidence that antibiotics prevent fever, swelling or problems with restricted mouth opening in patients who have had wisdom teeth removed.

There was no evidence to judge the effects of preventative antibiotics for extractions of severely decayed teeth, teeth in diseased gums, or extractions in patients who are sick or have low immunity to infection. Undertaking research in these groups of people may not be possible or ethical. However, it is likely that in situations where patients are at a higher risk of infection that preventative antibiotics may be beneficial, because infections in this group are likely to be more frequent and more difficult to treat.

Another concern, which cannot be assessed by clinical trials, is that widespread use of antibiotics by people who do not have an infection is likely to contribute to the development of bacterial resistance.

The conclusion of this review is that antibiotics given to healthy people to prevent infections, may cause more harm than benefit to both the individual patients and the population as a whole.

# SUMMARY OF FINDINGS FOR THE MAIN COMPARISON [Explanation]

Antibiotic compared to placebo for preventing infectious complications after tooth extraction

Patient or population: Patients undergoing tooth extraction Settings: Oral surgery referral centre Intervention: Antibiotic Comparison: Placebo

Outcomes	Illustrative comparative risks* (95% CI)		Relative effect (95% Cl)	No of participants (studies)	Quality of the evidence Comments (GRADE)
	Assumed risk	Corresponding risk			
	Placebo	Antibiotic			
<b>Local sign of infection</b> Follow-up: mean 7 days	118 per 1000	<b>34 per 1000</b> (19 to 59)	<b>RR 0.29</b> (0.16 to 0.50)	1523 (10 studies)	$\oplus \oplus \oplus \bigcirc$ moderate <sup>1</sup>
<b>Dry socket</b> Follow-up: mean 7 days	69 per 1000	<b>43 per 1000</b> (28 to 65)	<b>RR 0.62</b> (0.41 to 0.95)	1429 (9 studies)	$\oplus \oplus \oplus \bigcirc$ moderate <sup>2</sup>
Pain (dichotomous on 6- 7th day) Follow-up: mean 6.5 days	126 per 1000	<b>76 per 1000</b> (40 to 140)	<b>RR 0.60</b> (0.32 to 1.11)	675 (3 studies)	⊕⊕⊖⊖ Iow <sup>3,4</sup>
Pain score (VAS 7th day) Scale from: 1 to 100. Follow-up: mean 7 days	Mean pain score (VAS) in placebo groups was 15	The mean pain score (VAS 7th day) in the in- tervention groups was <b>8.</b> <b>17 lower</b> (11.9 to 4.45 lower)		372 (4 studies)	⊕⊕⊕⊖ moderate⁵
<b>Fever (6th-7th day)</b> Follow-up: mean 6.5 days	39 per 1000	<b>13 per 1000</b> (2 to 78)	<b>RR 0.34</b> (0.06 to 1.99)	816 (4 studies)	⊕⊕⊖⊖ Iow <sup>6,7</sup>
<b>Swelling (7th day)</b> Follow-up: mean 7 days	307 per 1000	<b>282 per 1000</b> (200 to 399)	<b>RR 0.92</b> (0.65 to 1.30)	334 (3 studies)	⊕⊕⊕⊖ moderate <sup>8</sup>

<ul> <li>*The basis for the assumed risk (e.g. the mean control group risk across studies) is provided in footnotes. The corresponding risk (and its 95% Cl) is based on the assumed risk in comparison group and the relative effect of the intervention (and its 95% Cl).</li> <li>Cf: confidence interval; RR: risk ratio.</li> <li>GRADE Working Group grades of evidence</li> <li>High quality: Eurther research is is very unlikely to change our confidence in the estimate of effect.</li> <li>Moderate quality: Further research is is very unlikely to have an important impact on our confidence in the estimate of effect and may change the estimate.</li> <li>Low quality: Further research is is very uncertain about the estimate on our confidence in the estimate of effect and is likely to change the estimate.</li> <li>Very low quality: We are very uncertain about the estimate.</li> <li>Very low quality: We are very uncertain about the estimate.</li> <li>Very low quality: We are very uncertain about the estimate.</li> <li>9 double-blind placebo-controlled trials - 7 trials at high risk of bias and 2 trials at unclear risk of bias.</li> <li>9 double-blind placebo-controlled trials - 7 trials at high risk of bias and 1 trial at unclear risk of bias.</li> <li>9 double-blind placebo-controlled trials - 7 trials at high risk of bias and 1 trial at unclear risk of bias.</li> <li>9 double-blind placebo-controlled trials - 3 at high risk of bias and 1 trial at unclear risk of bias.</li> <li>9 double-blind placebo-controlled trials - 3 at high risk of bias.</li> <li>9 double-blind placebo-controlled trials - 4 at high risk of bias and 1 at unclear risk of bias.</li> <li>9 double-blind placebo-controlled trials - 4 at high risk of bias and 1 at unclear risk of bias.</li> <li>9 double-blind placebo-controlled trials - 4 at high risk of bias and 1 at unclear risk of bias.</li> </ul>	Adverse effects Follow-up: mean 7 days	49 per 1000	<b>96 per 1000</b> (54 to 175)	<b>RR 1.98</b> (1.10 to 3.59)	930 (5 studies)	⊕⊕⊕⊖ moderate <sup>9</sup>	
<ul> <li>GRADE Working Group grades of evidence</li> <li>High quality: Further research is very unlikely to have an important impact on our confidence in the estimate of effect and may change the estimate.</li> <li>Low quality: Further research is very likely to have an important impact on our confidence in the estimate of effect and is likely to change the estimate.</li> <li>Very low quality: We are very uncertain about the estimate.</li> <li>10 double-blind placebo-controlled trials - 7 trials at high risk of bias and 3 trials at unclear risk of bias.</li> <li>9 double-blind placebo-controlled trials - 7 trials at high risk of bias and 1 trial at unclear risk of bias.</li> <li>3 double-blind placebo-controlled trials - 2 trials at high risk of bias and 1 trial at unclear risk of bias.</li> <li>4 double-blind placebo-controlled trials - 3 at high risk of bias.</li> <li>4 double-blind placebo-controlled trials - 3 at high risk of bias and 1 at unclear risk of bias.</li> <li>Substantial heterogeneity I<sup>2</sup> 57% P = 0.07.</li> <li>4 double-blind placebo-controlled trials - 3 at high risk of bias.</li> <li>5 double-blind placebo-controlled trials - 4 at high risk of bias.</li> <li>5 double-blind placebo-controlled trials - 4 at high risk of bias and 1 at unclear risk of bias.</li> </ul>	*The basis for the <b>assum</b> comparison group and the <b>Cl:</b> confidence interval; <b>Ri</b>	ed risk (e.g. the mea e relative effect of th R: risk ratio.	an control group risk across e intervention (and its 95% (	s studies) is provided in foc CI).	otnotes. The <b>correspondi</b>	<b>ng risk</b> (and its 95% CI) is based o	n the assumed risk in th
<ul> <li>Woolerable updative: Further research is likely to have an important impact on our confidence in the estimate of effect and its likely to change the estimate.</li> <li>Low quality: Further research is very likely to have an important impact on our confidence in the estimate of effect and its likely to change the estimate.</li> <li>10 double-blind placebo-controlled trials - 7 trials at high risk of bias and 3 trials at unclear risk of bias.</li> <li>9 double-blind placebo-controlled trials - 2 trials at high risk of bias and 1 trial at unclear risk of bias.</li> <li>Substantial heterogeneity I<sup>2</sup> 57% P = 0.07.</li> <li>4 double-blind placebo-controlled trials - 3 at high risk of bias.</li> <li>4 double-blind placebo-controlled trials - 3 at high risk of bias and 1 at unclear risk of bias.</li> <li>Substantial heterogeneity I<sup>2</sup> 60% (P = 0.11).</li> <li>3 double-blind placebo-controlled trials - 4 at high risk of bias.</li> <li>5 double-blind placebo-controlled trials - 4 at high risk of bias and 1 at unclear risk of bias.</li> </ul>	GRADE Working Group gr High quality: Further rese	ades of evidence arch is very unlikely t	o change our confidence in	the estimate of effect.			
<ul> <li>10 double-blind placebo-controlled trials - 7 trials at high risk of bias and 3 trials at unclear risk of bias.</li> <li>9 double-blind placebo-controlled trials - 2 trials at high risk of bias and 1 trial at unclear risk of bias.</li> <li>3 double-blind placebo-controlled trials - 2 trials at high risk of bias and 1 trial at unclear risk of bias.</li> <li>Substantial heterogeneity I<sup>2</sup> 57% P = 0.07.</li> <li>4 double-blind placebo-controlled trials - 3 at high risk of bias.</li> <li>4 double-blind placebo-controlled trials - 3 at high risk of bias.</li> <li>4 double-blind placebo-controlled trials - 3 at high risk of bias.</li> <li>4 double-blind placebo-controlled trials - 3 at high risk of bias.</li> <li>5 double-blind placebo-controlled trials - 4 at high risk of bias.</li> <li>5 double-blind placebo-controlled trials - 4 at high risk of bias and 1 at unclear risk of bias.</li> </ul>	Moderate quality: Further Low quality: Further resea Very low quality: We are	research is likely to arch is very likely to h very uncertain about	have an important impact of have an important impact on the estimate.	our confidence in the estin	nate of effect and may ch nate of effect and is likely	ange the estimate. to change the estimate.	
Substantial heterogeneity $l^2$ 57% P = 0.07. 4 double-blind placebo-controlled trials - all at high risk of bias. 4 double-blind placebo-controlled trials - 3 at high risk of bias and 1 at unclear risk of bias. Substantial heterogeneity $l^2$ 60% (P = 0.11). 3 double-blind placebo-controlled trials - 4 at high risk of bias and 1 at unclear risk of bias. 5 double-blind placebo-controlled trials - 4 at high risk of bias and 1 at unclear risk of bias.	10 double-blind placebo- 9 double-blind placebo-c	controlled trials - 7 tr ontrolled trials - 7 tria	ials at high risk of bias and als at high risk of bias and 2	3 trials at unclear risk of bia trials at unclear risk of bias	IS.		
<ul> <li>4 double-blind placebo-controlled trials - 3 at high risk of bias and 1 at unclear risk of bias.</li> <li>Substantial heterogeneity l<sup>2</sup> 60% (P = 0.11).</li> <li>3 double-blind placebo-controlled trials at high risk of bias.</li> <li>5 double-blind placebo-controlled trials - 4 at high risk of bias and 1 at unclear risk of bias.</li> </ul>	3 double-blind placebo-c Substantial heterogeneity 4 double-blind placebo-c	ontrolled trials - 2 trials $P^{2}$ 57% P = 0.07. ontrolled trials - all at	high risk of bias.	that at unclear risk of blas.			
5 double-blind placebo-controlled trials - 4 at high risk of bias and 1 at unclear risk of bias.	4 double-blind placebo-c Substantial heterogeneity 3 double-blind placebo-c	ontrolled trials - 3 at $I^2$ 60% (P = 0.11).	high risk of bias and 1 at un n risk of bias.	clear risk of bias.			
	5 double-blind placebo-c	ontrolled trials - 4 at	high risk of bias and 1 at un	clear risk of bias.			

# BACKGROUND

#### **Description of the condition**

Tooth extraction is a very common surgical procedure, and is most frequently done by general dental practitioners. In spite of the steady decrease in routine extraction of permanent teeth registered in the last decades (Thomas 1994; Sleeman 1995; McCaul 2001), general dental practitioners from European countries may extract up to seven teeth per week (McCaul 2001). An estimated 17% of patients undergo extractions over a 5-year period (Worthington 1999), with the highest tooth extraction rate per patient being among patients in the sixth and seventh decade of life (Chrysanthakopoulos 2011). The main reasons for extraction of permanent teeth are still caries and periodontal disease, in variable proportions according to age of patients, country and year of publication (Additional Table 1). Wisdom teeth failing to erupt or erupting only partially represent a distinct category of dental elements named impacted (third molar) teeth. In fact, impacted wisdom teeth are extracted either because of local inflammatory problems, or in order to avoid possible future complications (although a recent Cochrane review did not find sufficient evidence to support or refute routine prophylactic removal of asymptomatic impacted wisdom teeth in adults (Mettes 2012)).

The main objective for a successful surgery is to minimise, as much as possible, patient discomfort in the post-operative period after tooth extraction. Symptoms such as pain, swelling, trismus, fever and dry socket are complications which are unpleasant for patients and could generate difficulty in chewing, in speaking, in performing oral hygiene, and alteration of other activities of daily living, resulting in days off from work or study. All these complications depend on inflammatory response, but they can be due to subsequent infection, for example if surgical trauma is in a contaminated area (where severe caries or periodontitis is present) or where more complex and aggressive procedures are performed (e.g. ostectomy).

Signs of post-extraction infectious complications include abscess, pain, fever, swelling, trismus. Another complication of putative bacterial origin is alveolar osteitis (dry socket), a painful condition which follows the dissolution of the blood clot which occurs as a result of bacterial invasion. The overall incidence of postoperative infections is relatively low (Jaafar 2000; Bouloux 2007; Bortoluzzi 2010), however antibiotics are frequently prescribed in a prophylactic way, particularly in case of complicated surgeries and patients with systemic conditions potentially causing immunodeficiency, such as HIV infection, diabetes and cancer (Epstein 2000). There are a range of antibiotics which are effective in treating dental infections. These include penicillin, amoxicillin, erythromycin, clindamycin, doxycycline and metronidazole which are usually administered orally, between one and four times daily. Alternatively antibiotics can also be administered by parenteral or local routes.

#### How the intervention might work

The oral environment contains a range of bacteria which have the potential to cause painful infections in wounds. Antibiotics are effective in treating such infections and are also likely to act to prevent the development of painful wound infections. The optimal timing of the dose or doses is unclear. Antibiotics could be administered as a large single dose prior to the extraction, or a course of antibiotics taken over the post-operative period, or some combination of these. Adverse effects, such as diarrhoea or allergy due to antibiotics are also possible.

#### Why it is important to do this review

In 2010, a systematic review showed that both long duration and multiple courses of antibiotics prescribed in general medical practice were consistently associated with the development of bacterial resistance to those antibiotics in that individual and that the greater the number of antibiotic courses prescribed, the higher the chance of resistant bacteria development (Costelloe 2010). Dental prescribing accounts for a significant proportion of total antibacterial prescribing in primary care (7% to 9%) (Dar-Odeh 2010; Karki 2011). In addition, antibiotics used in dental practice can cause potentially serious adverse drug reactions and interactions (Hersh 1999). According to the European Commission, overuse and misuse of antibiotics are the main causes of microbial resistance to drugs. For this reason in 2011 an action plan to tackle microbial resistance to drugs was presented; the first aim of such plan is to make sure that antimicrobials are used appropriately both in humans and animals. Better evidence is needed about the use of antibiotic prophylaxis in patients undergoing tooth extraction in order to determine appropriate use (EU Commission 2011).

This systematic review will summarise the evidence of the effects of systemic antibiotics prescribed to prevent infectious complications following tooth extraction. A separate Cochrane systematic review evaluating interventions to manage dry socket following tooth extraction will be published in 2012 (Daly 2008).

# OBJECTIVES

#### **Description of the intervention**

• To assess the effects of antibiotic prophylaxis on the incidence of infectious complications following tooth extraction.

• To assess the effects of antibiotic prophylaxis following tooth extraction in immunosuppressed patients (e.g. HIV infection, AIDS, diabetes, transplants) or patients with other conditions (e.g. bone diseases).

• To assess the effects of antibiotic prophylaxis in particular procedures, such as extraction of impacted teeth or wisdom teeth.

# METHODS

#### Criteria for considering studies for this review

#### **Types of studies**

Randomised controlled trials with a double-blind design (participants and assessors) were included. Cross-over studies were included providing the interval (or washout period) between interventions was at least 6 weeks.

#### **Types of participants**

Anyone undergoing a tooth extraction, including extraction of impacted teeth.

#### **Types of interventions**

#### Active

1) Any regimen of systemic antibiotic prophylaxis (i.e. prescribed in absence of infection) administered before or after tooth extraction. Topical antibiotic therapy was not included.

#### Control

1) Placebo.

#### Types of outcome measures

#### **Primary outcomes**

• Post-surgical complications of putative infectious nature, including: alveolar osteitis (dry socket), pain, fever, swelling, trismus.

#### Secondary outcomes

- Other post-surgical complications.
- Any adverse effect related to antibiotics.

Trials which reported the outcomes of endocarditis incidence, bacteriaemia or serum markers of infection only, were not considered for inclusion in this review.

#### Search methods for identification of studies

For the identification of studies included or considered for this review, we developed detailed search strategies for each database searched. These were based on the search strategy developed for MEDLINE (OVID) but revised appropriately for each database. The search strategy used a combination of controlled vocabulary and free text terms and was linked with the Cochrane Highly Sensitive Search Strategy (CHSSS) for identifying randomised trials (RCTs) in MEDLINE: sensitivity maximising version (2008 revision) as referenced in Chapter 6.4.11.1 and detailed in box 6.4.c of the *Cochrane Handbook for Systematic Reviews of Interventions* version 5.1.0 (updated March 2011) (Higgins 2011).

We searched the following electronic databases:

• The Cochrane Oral Health Group's Trials Register (to 25 January 2012) (Appendix 1)

Cochrane Central Register of Controlled Trials

- (CENTRAL) (*The Cochrane Library* 2012, Issue 1) (Appendix 2)
- MEDLINE via OVID (1948 to 25 January 2012) (Appendix 3)
- EMBASE via OVID (1980 to 25 January 2012) (Appendix 4)

• LILACS via BIREME (1982 to 25 January 2012) (Appendix 5).

Details of the MEDLINE search are provided in Appendix 3. We linked the search of EMBASE to the Cochrane Oral Health Group filter for identifying RCTs (Appendix 4), and the search of LILACS to the Brazilian Cochrane Center filter (Appendix 5).

#### Handsearching

The following journals were identified as being important to handsearch for this review:

- British Journal of Oral and Maxillofacial Surgery
- British Dental Journal
- International Journal of Oral and Maxillofacial Surgery
- Journal of the American Dental Association
- Journal of Dental Research
- Journal of Dentistry

 Oral Surgery, Oral Pathology, Oral Medicine, Oral Radiology and Endodontics.

The handsearching was done as part of the Cochrane Worldwide Handsearching Programme. *See* the Cochrane Masterlist of Journals for details of the volumes and issues that have been searched to date.

Reference lists of all eligible trials and of existing reviews were checked for additional studies. The first named authors of all included trials were contacted in an attempt to identify unpublished studies and to obtain further information about the trials.

There were no restrictions on language or date of publication.

### Data collection and analysis

#### Selection of studies

The title and abstract of each article resulting from the different search strategies were examined independently by two review authors (Giovanni Lodi (GL) and Susan Furness (SF)). Where studies appeared to meet the inclusion criteria for this review or where there were insufficient data in the title and abstract to make a clear decision, the full report was obtained. The full report was then assessed by at least two review authors to determine whether studies met the inclusion criteria. Any disagreements were resolved by discussion. Studies rejected at this or subsequent stages were recorded in the Characteristics of excluded studies table where the reason(s) for exclusion were recorded. A flow chart to summarise the results of the search was prepared (Figure 1).



Antibiotics to prevent complications following tooth extractions (Review) Copyright © 2013 The Cochrane Collaboration. Published by John Wiley & Sons, Ltd.

#### Data extraction and management

All studies which met the inclusion criteria for this review underwent risk of bias assessment and data extraction using a specially designed data extraction form. Data were extracted by at least two review authors independently and were also entered into a spreadsheet. Any disagreement was discussed and agreement reached. When necessary authors were contacted for clarification or missing information.

For each trial the following data were recorded.

• Year of publication, country of origin, number of centres, source of study funding, recruitment period.

• Details of the participants including demographic characteristics and criteria for inclusion and exclusion, type of teeth being extracted and reasons, numbers randomised to each treatment group.

• Details of the type of antibiotic, dose, mode of administration, time of administration relative to the extraction procedure and duration of antibiotic treatment.

• Details of other concomitant treatments - type of anaesthetic, mouthrinses, pain management.

- Details of the outcomes reported, including method of assessment, and time(s) assessed.
  - Description of operators.
  - Sample size calculation.

#### Assessment of risk of bias in included studies

Each of the trials included in this review was assessed for risk of bias using The Cochrane Collaboration's risk of bias assessment tool (Higgins 2011). The following six domains were assessed for each trial: random sequence generation, allocation concealment, blinding, completeness of outcome data, risk of selective outcome reporting and risk of other bias.

A description of what was reported to have occurred was included for each domain in each trial, together with a judgement of low, unclear or high risk of bias. For example criteria, used for risk of bias judgements for allocation concealment as described in the *Cochrane Handbook for Systematic Reviews of Interventions* (Higgins 2011) are described below.

1) Method of allocation concealment. Criteria for the judgement of low risk of bias: when the randomisation schedule is concealed from the researcher recruiting participants to the trial by means of either:

• central allocation (including telephone, web-based and pharmacy-controlled randomisations); or

• sequentially numbered drug containers of identical appearance;or

• sequentially numbered, opaque, sealed envelopes.

When this information is not reported, the domain was judged as unclear.

2) Protection against performance and detection bias (blindness of the study). One of the inclusion criteria for this review is that trials be double blind. This is interpreted as meaning that neither the participants nor the researchers assessing the outcomes of the trial were aware of the allocated treatment, unless further information is given.

3) Incomplete outcome data. Criteria for the judgement of low risk of bias for this domain are:

• no missing outcome data; or

• less than 20% of randomised participants excluded from the analysis and numbers of trial participants excluded balanced in numbers across intervention groups, with similar reasons for missing data across groups;

#### and

• for dichotomous outcome data, the proportion of missing outcomes compared with observed event rate not great enough to have a clinically relevant impact on the effect estimate.

#### Overall risk of bias

A summary assessment of the risk of bias was undertaken (Higgins 2011) as follows:

1) low risk of bias: all of the domains judged to be at low risk of bias

2) unclear risk of bias: one or more domains judged to be at unclear risk of bias

3) high risk of bias: one or more domains judged to be at high risk of bias.

Risk of bias assessment of the studies was carried out without blinding the name of authors, institutions and journal. Data about the study, its eligibility, validity, design and outcome information, were recorded by each review author on an extraction form. In case of disagreement, consensus was achieved by discussion.

#### Measures of treatment effect

The primary measure of intervention effect was reduction in incidence of infectious complications, such as alveolar osteitis (dry socket), pain, fever, swelling or trismus between the control and intervention group. Dichotomous data are expected for these. Other dichotomous data may include the incidence of adverse effects. For each intervention, data on the number of patients of intervention and control group who experienced the event (outcome) and the total number of patients, were sought and summarised. Dichotomous data were analysed by calculating risk ratios and 95% confidence intervals.

Where pooling of data from both parallel and cross-over studies was appropriate we used the generic inverse variance method to enter the data into Review Manager (RevMan) software (Higgins 2011).

#### Dealing with missing data

Missing data were obtained from tables and graphs if possible. Where data were missing or unclear we attempted to contact the authors of the studies to request clarification or additional data.

#### Assessment of heterogeneity

We assessed heterogeneity by inspection of the point estimates and confidence intervals on the forest plots. We assessed the variation in treatment effects by means of Cochran's test for heterogeneity and quantified by the  $I^2$  statistic. We considered heterogeneity statistically significant if the P value was < 0.1. A rough guide to the interpretation of the  $I^2$  statistic given in the *Cochrane Handbook for Systematic Reviews of Interventions* is: 0% to 40% might not be important, 30% to 60% may represent moderate heterogeneity, 50% to 90% may represent substantial heterogeneity, 75% to 100% considerable heterogeneity (Higgins 2011).

#### Assessment of reporting biases

Only a proportion of research projects conducted are ultimately published in an indexed journal and become easily identifiable for inclusion in systematic reviews. Reporting biases arise when the reporting of research findings is influenced by the nature and direction of the findings of the research. We attempted to minimise potential reporting biases including publication bias, time lag bias, multiple (duplicate) publication bias and language bias in this review.

If there had been more than ten studies in one outcome we planned to construct a funnel plot. If there were asymmetry in the funnel plot indicating possible publication bias we planned to undertake statistical analysis using the methods introduced by Egger 1997 (continuous outcome) and Rücker 2008 (dichotomous outcome). We attempted to avoid time lag bias, multiple (duplicate) publication bias and language bias by conducting a detailed sensitive search, including searching for ongoing studies. There were no restrictions on language, and we found translators for potentially relevant trials published in other languages.

#### Data synthesis

We only conducted a meta-analysis if there were studies of similar comparisons reporting the same outcome measures. We combined risk ratios for dichotomous data, and mean differences for continuous data, using random-effects models provided there were more than three studies in the meta-analysis. We combined the treatment effects from cross-over trials with those from parallel group trials where appropriate, using the data from both periods of the cross-over studies (Elbourne 2002). We used the generic inverse variance method incorporated in RevMan for all analyses that included cross-over trials using appropriate methods as outlined in the *Cochrane Handbook for Systematic Reviews of Interventions* (Elbourne 2002; Higgins 2011).

#### Subgroup analysis and investigation of heterogeneity

Whenever possible, subgroup analyses were undertaken based on time of administration (pre- or post-procedure) and the presence or absence of patients with systemic conditions (HIV, diabetes, etc).

#### Sensitivity analysis

A sensitivity analysis was undertaken, excluding studies at high risk of bias.

#### Presentation of main results

A summary of findings table was developed for the main outcomes of this review using GRADEPro software. We used the mean risk in the placebo groups of the included studies as the assumed risk for each outcome and calculated the corresponding risk using the risk ratio (or mean difference) estimate obtained from the metaanalysis. The quality of the body of evidence was assessed with reference to the overall risk of bias of the included studies, the directness of the evidence, the inconsistency of the results, the precision of the estimates, the risk of publication bias and the magnitude of the effect. The quality of the body of evidence for each of the main outcomes was categorised as high, moderate, low or very low.

# RESULTS

#### **Description of studies**

#### **Results of the search**

A total of 1053 references were identified by the electronic searches. The titles and abstracts were scanned by two review authors (Susan Furness (SF) and Giovanni Lodi (GL)) and 997 references were excluded as not relevant to this review. Full text copies of 56 potentially eligible papers were retrieved and after close reading, 37 studies were excluded (*see* Characteristics of excluded studies table). Two further studies were identified from searches of reference lists of included studies. Finally 21 references to 18 studies met the inclusion criteria for this review (Figure 1).

#### **Included studies**

#### Characteristics of trial design and setting

For a summary of the characteristics of each of the included studies *see* Characteristics of included studies table.

Of the 18 included studies, three were conducted in Spain (Arteagoitia 2005; Lacasa 2007; López-Cedrún 2011), three in Sweden (Bystedt 1980; Bystedt 1981; Bergdahl 2004), three in the UK (MacGregor 1980; Kaziro 1984; Mitchell 1986), two in India (Pasupathy 2011; Sekhar 2001) and one study was conducted in each of Brazil (Bezerra 2011), Colombia (Leon Arcila 2001), Denmark (Ritzau 1992), Finland (Happonen 1990), Poland (Kaczmarzyk 2007), New Zealand (Barclay 1987), and United States of America (Halpern 2007).

Seventeen studies used parallel group designs and one was a splitmouth cross-over trial (Bezerra 2011) where participants each had two extraction procedures, which were separated by a period of at least 45 days. Nine studies had two treatment arms (MacGregor 1980; Mitchell 1986; Barclay 1987; Ritzau 1992; Leon Arcila 2001; Bergdahl 2004; Arteagoitia 2005; Halpern 2007; Bezerra 2011), eight studies had three treatment arms (Bystedt 1981; Kaziro 1984; Happonen 1990; Sekhar 2001; Kaczmarzyk 2007; Lacasa 2007; López-Cedrún 2011; Pasupathy 2011) and one study (Bystedt 1980) had three subtrials, each with two or three arms. The data from these separately randomised subtrials were then combined and were unsuitable for inclusion in meta-analysis.

#### **Characteristics of participants**

The 18 included studies randomised more than 2456 participants to either an antibiotic or placebo. (One of the included trials ( MacGregor 1980) used an unusual design and did not state exactly how many participants were randomised and analysed.) All of the included studies compared at least one antibiotic regimen with placebo in patients undergoing dental extraction. Fifteen trials described extraction procedures done using local anaesthesia, two trials used general anaesthesia (MacGregor 1980; Halpern 2007), and in one trial the method of anaesthesia was not stated (Mitchell 1986).

In all of the included studies the participants underwent extraction of third molars only, and in the majority of the studies only mandibular third molars were included (Bystedt 1980; MacGregor 1980; Bystedt 1981; Mitchell 1986; Barclay 1987; Happonen 1990; Ritzau 1992; Sekhar 2001; Bergdahl 2004; Arteagoitia 2005; Kaczmarzyk 2007; Lacasa 2007; López-Cedrún 2011; Pasupathy 2011). Fourteen studies included participants with impacted teeth only (Bystedt 1980; MacGregor 1980; Bystedt 1981; Kaziro 1984; Mitchell 1986; Barclay 1987; Happonen 1990; Leon Arcila 2001; Sekhar 2001; Arteagoitia 2005; Kaczmarzyk 2007; Halpern 2007; Bezerra 2011; Pasupathy 2011), two studies included participants with either impacted or partially impacted teeth (Ritzau 1992; López-Cedrún 2011), one study included participants with only partially impacted teeth (Bergdahl 2004), and one study included participants with "teeth needing surgical extraction" (Lacasa 2007).

In one trial (Barclay 1987) the participants had a history of nonacute pericoronitis, and in another (Bergdahl 2004) 41% of participants had pericoronitis at some stage and were entered into the trial "after objective and subjective symptoms of pericoronitis had ceased", thus participants of both these studies were likely to be at higher risk of infectious complications. Recent episodes of local infection were reason for exclusion in two other studies (Sekhar 2001; Lacasa 2007). In the remaining trials, participants were considered healthy at baseline and systemic conditions, including those causing immunosuppression, were often reason for exclusion from the trial (*see* Characteristics of included studies). None of the trials assessed the effect of antibiotic prophylaxis in

patients who required extraction of one or more teeth due to caries or periodontal disease, even though these indications are the most common reasons for tooth extraction.

#### **Characteristics of interventions**

In 16 trials the antibiotics were administered orally, one study ( Halpern 2007) used intravenous penicillin or clindamycin and one study (MacGregor 1980) administered penicillin intramuscularly. The antibiotic interventions were classified into three groups, based on the time of administration relative to the extraction (studies with three or more arms may be included in more than one group).

• Antibiotics given pre-operatively only (30 minutes to 2 hours prior to procedure): MacGregor 1980; Mitchell 1986; Ritzau 1992; Sekhar 2001; Bergdahl 2004; Halpern 2007; Kaczmarzyk 2007; Lacasa 2007; Bezerra 2011; López-Cedrún 2011.

• Antibiotics given post-operatively only: Kaziro 1984; Sekhar 2001; Arteagoitia 2005; López-Cedrún 2011.

• Antibiotics given both pre- and post-operatively: Bystedt 1980; Bystedt 1981; Barclay 1987; Happonen 1990; Leon Arcila 2001; Kaczmarzyk 2007; Lacasa 2007.

The antibiotics selected for use in the studies were amoxicillin (Leon Arcila 2001; Bezerra 2011; López-Cedrún 2011; Pasupathy 2011), a combination of amoxicillin/clavulanate ( Arteagoitia 2005; Lacasa 2007), azidocillin (Bystedt 1980; Bystedt 1981), clindamycin (Bystedt 1980; Halpern 2007; Kaczmarzyk 2007), doxycycline (Bystedt 1980), erythromycin (Bystedt 1980), metronidazole (Kaziro 1984; Barclay 1987; Ritzau 1992; Sekhar 2001; Bergdahl 2004; Pasupathy 2011), penicillin (MacGregor 1980; Halpern 2007), phenoxymethylpenicillin (Happonen 1990) and tinidazole (Mitchell 1986; Happonen 1990). Details of specific dosage regimens are recorded in the Characteristics of included studies for each study.

#### **Characteristics of outcomes**

Four studies (Bystedt 1981; Ritzau 1992; Bergdahl 2004; Arteagoitia 2005) reported the development of dry socket. Ritzau 1992 with a follow-up at 7 days, Arteagoitia 2005 with a followup at 7 days and 8 weeks, Bergdahl 2004 with a follow-up between 2 and 4 days after surgery, and Bystedt 1981 with a follow-up at 2, 5 and 7 days.

Five studies (Bystedt 1981; Happonen 1990; Arteagoitia 2005; Kaczmarzyk 2007; Lacasa 2007) investigated pain. Arteagoitia 2005 with a follow-up at 48 hours and 6 days, Bystedt 1981 with a follow-up at 2, 5 and 7 days, Happonen 1990 had a follow-up at 6 days, Kaczmarzyk 2007 with a follow-up at 1, 2 and 7 days and Lacasa 2007 with a follow-up at 1, 3 and 7 days.

Fever in the first 24 hours was evaluated by Bystedt 1981; Happonen 1990; Arteagoitia 2005; Kaczmarzyk 2007 and Lacasa 2007, and three studies (Bystedt 1981; Kaczmarzyk 2007; Lacasa 2007) included swelling among outcomes.

Four studies investigated trismus among outcomes (Bystedt 1981; Happonen 1990; Kaczmarzyk 2007; Lacasa 2007), where Happonen 1990 had a follow-up at 6 days.

Bystedt 1981 reported that there were no non-infectious complications following extraction, and Kaczmarzyk 2007 stated that gastric complications in the antibiotic group following extraction was the reason that 3 of the 100 participants in this trial were excluded from the analysis. The other studies did not report other complications following extraction.

Adverse effects were reported per participant by only 5 of the 18 trials included in this review (Bystedt 1981; Barclay 1987; Arteagoitia 2005; Kaczmarzyk 2007; Lacasa 2007).

#### **Excluded studies**

For the main reason for excluding each study *see* Characteristics of excluded studies table.

A total of 37 studies were listed as excluded from this review after the full text of the paper was reviewed by two or more authors. Twenty-two studies because they were not double blind (Curran 1974; Krekmanov 1980; Krekmanov 1981; Krekmanov 1986; Lombardia Garcia 1987; Mitchell 1987; Abu-Mowais 1990; Lyall 1991; Samsudin 1994; Walkow 1995; Monaco 1999; Yoshii 2002; Delilbasi 2004; Foy 2004; Poeschl 2004; Graziani 2005; Sulejmanagie 2005; Uluibau 2005; Grossi 2007; Ataoglu 2008; Monaco 2009; Lopes 2011). Four studies were excluded because two antibiotic regimens were compared directly with no placebocontrolled group (Laird 1972; Limeres 2009; Luaces-Rey 2010; Olusanya 2011) and three trials were excluded because interventions were not randomly allocated (Osborn 1979; Rood 1979; Fridrich 1990). Two cross-over trials were excluded because the washout period between interventions was less than 6 weeks (Siddigi 2010; de Moura 2011) and four trials because they evaluated topical antibiotics only (MacGregor 1973; Swanson 1989; Reekie 2006; Stavropoulos 2006). One trial was excluded because it evaluated antibiotics in conjunction with a range of dental surgical procedures not just extractions (Bargnesi 1985) and one because it presented data on bacteraemia outcomes only (Head 1984).

#### Risk of bias in included studies

#### Allocation

#### Sequence generation

Nine studies were assessed as being at low risk of bias for this domain. Four studies reported that randomisation was generated by computer (Ritzau 1992; Leon Arcila 2001; Arteagoitia 2005; Pasupathy 2011), two studies used random number tables (Barclay 1987; Kaczmarzyk 2007), two used predetermined random codes (Mitchell 1986; López-Cedrún 2011) and one used a coin toss (Bezerra 2011). The remaining nine studies gave no details about the method of sequence generation and were assessed at unclear risk of bias for this domain (Bystedt 1980; MacGregor 1980; Bystedt 1981; Kaziro 1984; Happonen 1990; Sekhar 2001; Bergdahl 2004; Halpern 2007; Lacasa 2007).

#### Allocation concealment

Eleven studies described adequate allocation concealment (Kaziro 1984; Mitchell 1986; Ritzau 1992; Leon Arcila 2001; Sekhar 2001; Arteagoitia 2005; Halpern 2007; Kaczmarzyk 2007; Bezerra 2011; López-Cedrún 2011; Pasupathy 2011) and were assessed at low risk of bias for this domain. For the remaining seven studies allocation concealment was not reported and these studies were assessed as at unclear risk of bias for this domain.

Overall eight trials were considered to be at low risk of selection bias (Mitchell 1986; Ritzau 1992; Leon Arcila 2001; Arteagoitia 2005; Kaczmarzyk 2007; Bezerra 2011; López-Cedrún 2011; Pasupathy 2011) and for the remaining 10 studies the risk of selection bias was unclear.

#### Blinding

The inclusion criteria for this review specified that trials be double blind. Where trials only reported 'double blind' with no further details we interpreted this as meaning that both the participant and the person who assessed the outcomes (either the surgeon or the patient) were blinded to the allocated treatment. Consequently all included trials were assessed as being at low risk of performance and detection bias.

#### Incomplete outcome data

Most of the included trials had relatively low rates of participants excluded from the analysis due to loss to follow-up or withdrawal from the trials. However, these trials also reported low event rates for the outcomes of interest, which meant that even small numbers of patients excluded could have introduced a bias.

Three trials reported that all the randomised participants were included in the analysis (Bystedt 1981; Mitchell 1986; Leon Arcila 2001) and in two trials attrition was less than 1% (Bergdahl 2004; Arteagoitia 2005). In the split-mouth study by Bezerra 2011, two participants were lost to follow-up but this was not considered to introduce a risk of attrition bias due to the study design. These six trials were assessed as being at low risk of attrition bias.

Three trials (Bystedt 1980; MacGregor 1980; Kaziro 1984) did not report the number of randomised participants included in the analysis and these trials were published more than 25 years ago and we were unable to obtain this information. In the study by Halpern 2007 there were more drop-outs in the placebo group than the antibiotic group and it was unclear whether this could have introduced a bias. These four trials were assessed as being at unclear risk of attrition bias.

Three trials (Barclay 1987; Sekhar 2001; López-Cedrún 2011) reported overall exclusion of participants from outcome evaluation of 10%, 8% and 17% respectively, and noted that losses were unequally distributed between antibiotic and placebo groups. A further five trials (Happonen 1990; Ritzau 1992; Kaczmarzyk 2007; Lacasa 2007; Pasupathy 2011) reported between 5% and 14% of participants were excluded from the outcome evaluation and did not describe the reasons or the treatment group from which participants were excluded. These eight trials were all assessed as being at high risk of attrition bias.

#### Selective reporting

Selective reporting is difficult to assess in the absence of a trial protocol. We based our assessment on three factors: whether the trial report contained in the results section, data on all the outcome measures described in the methods section of the report; whether planned outcome measures included those that would reasonably be expected in such a trial; and whether both point estimates and variances were reported.

Five trials (Barclay 1987; Sekhar 2001; Kaczmarzyk 2007; Bezerra 2011; López-Cedrún 2011) reported complete data on all the outcomes that were listed in their methods sections and were consequently assessed as being at low risk of reporting bias.

The authors in Bystedt 1981 evaluated swelling and trismus but did not report these data, stating only that there was no difference between the groups. In Kaziro 1984 planned outcomes were reported only in graphical form as percentages in each group. A further four trials (Mitchell 1986; Ritzau 1992; Leon Arcila 2001; Halpern 2007) reported the planned single outcome but did not report pain, swelling or trismus which we consider to be important outcomes following this procedure. These six trials were assessed as being at unclear risk of reporting bias.

The remaining seven trials (Bystedt 1980; MacGregor 1980; Happonen 1990; Bergdahl 2004; Arteagoitia 2005; Lacasa 2007; Pasupathy 2011) were assessed as being at high risk of reporting bias. The trial by Arteagoitia 2005 planned to measure pain scores on a VAS but these were not reported. Bergdahl 2004 did not report data on three of the outcomes listed in the methods of the report, and Lacasa 2007 listed eight planned outcomes of which one was reported fully, one (pain) was reported as a mean for each group without an estimate of variance and the remaining six outcomes were not reported at all. Bystedt 1980 reported planned outcomes as point estimates without estimates of variance, but not for each group to which participants were randomised in the three subtrials in this report, and MacGregor 1980 reported pain, swelling, and trismus for all participants combined, in graphical form only, making it impossible to determine the effects of the interventions on these outcomes. Happonen 1990 did not report the pain measured by VAS at 7 days despite stating that pain was the main reason given for participants to be unable to work. Pasupathy 2011 did not report the outcomes of pain swelling or trismus but it would appear that these data were collected.

#### Other potential sources of bias

For nine of the included studies there were no other sources of bias identified (MacGregor 1980; Bystedt 1981; Mitchell 1986; Barclay 1987; Happonen 1990; Leon Arcila 2001; Arteagoitia 2005; Kaczmarzyk 2007; Halpern 2007).

Two trials were assessed as at high risk of other bias (Bezerra 2011; Pasupathy 2011). The trial by Bezerra 2011 was a splitmouth study in which participants each underwent two extraction procedures a minimum of 45 days apart. The trial report states that "the mean pain scores were lower on the last assessment [day 14] compared with the first [baseline] in both groups" suggesting that there may have been a period effect. The power calculation reported in Pasupathy 2011 suggests that this trial is underpowered which is likely to bias results towards the null hypothesis of no difference between antibiotic and placebo.

In the remaining seven trials risk of other bias was unclear.

#### **Overall risk of bias**

None of the trials included in this review were assessed as at low risk of bias for all the domains (Figure 2; Figure 3). Five trials (Bystedt 1981; Kaziro 1984; Mitchell 1986; Leon Arcila 2001; Halpern 2007) were assessed as at unclear risk of bias because there was insufficient information in the trial report or available from the authors to determine risk of bias in at least one domain. The remaining 13 trials (Bystedt 1980; MacGregor 1980; Barclay 1987; Happonen 1990; Ritzau 1992; Sekhar 2001; Bergdahl 2004; Arteagoitia 2005; Kaczmarzyk 2007; Lacasa 2007; Bezerra

2011; López-Cedrún 2011; Pasupathy 2011) were assessed as at high overall risk of bias because each of these trials was at high risk of bias in one or more domains.

#### Blinding of participants and personnel (performance bias) Blinding of outcome assessment (detection bias) Random sequence generation (selection bias) Allocation concealment (selection bias) Incomplete outcome data (attrition bias) Selective reporting (reporting bias) Other bias Đ Đ Ŧ Arteagoitia 2005 Barclay 1987 ? Đ æ 64 Bergdahl 2004 ? ? Đ ? Ŧ Bezerra 2011 Đ Ŧ Đ Bystedt 1980 ? ? ? ? Đ (Ŧ Bystedt 1981 ? ? ? Đ Đ Đ Đ Halpern 2007 ? Đ Đ ? ? Happonen 1990 ? ? Đ (Ŧ Kaczmarzyk 2007 Đ Đ Đ Đ Ŧ Ŧ Kaziro 1984 Ŧ ? ? ? ? Ŧ Lacasa 2007 ? ? ? Đ Ŧ Leon Arcila 2001 ? Đ Đ Đ Đ Đ Đ López-Cedrún 2011 Ŧ ? Đ Ŧ MacGregor 1980 ? ? Đ ? Ŧ (Ŧ Mitchell 1986 ? æ Đ Đ Đ Ŧ **(+** Pasupathy 2011 Ŧ Ŧ Ritzau 1992 ? ? + (Ŧ Sekhar 2001 ?

# Figure 2. Risk of bias summary: review authors' judgements about each risk of bias item for each included study.

Antibiotics to prevent complications following tooth extractions (Review)

Copyright © 2013 The Cochrane Collaboration. Published by John Wiley & Sons, Ltd.





#### **Effects of interventions**

See: **Summary of findings for the main comparison** Antibiotic compared to placebo for preventing infectious complications after tooth extraction

Three of the trials which met the inclusion criteria for this review did not report data in a form that was suitable for inclusion in meta-analysis (Bystedt 1980; MacGregor 1980; Kaziro 1984). The trial by MacGregor 1980 compared a single dose of intramuscular penicillin with placebo, followed up participants for 4 days and reported only that there were no significant differences between antibiotic and placebo with regard to pain, swelling and trismus but provided no data to substantiate this claim. The authors of Bystedt 1980 conducted three independent subtrials but reported data combining all of these trials. We were unable to draw any conclusions based on the data presented in this paper. In Kaziro 1984 the authors did not report the number of participants included in the outcome assessments but used graphs to report the percentage of participants with infections, pain and swelling. There were fewer patients in the antibiotic group who reported infections or pain but there were no estimates of variance so the statistical significance (if any) cannot be determined from this report (Additional Table 2).

The results from the remaining 15 trials are described below in subgroups depending on the time(s) the antibiotics were administered (either pre-operatively, post-operatively or both pre- and post-operatively). Subgroup analysis based on the immune state of patients was not possible, as studies on immunosuppressed patients, or those with underlying health conditions which may have influenced their immune system, were not identified by our searches. Where there were few trials reporting an outcome, the subgroups were indicated by footnotes but in these cases separate subgroup estimates were not reported (Analysis 1.5; Analysis 1.7; Analysis 1.8).

#### Local sign of infection

#### **Pre-operative antibiotics**

Seven trials reported the outcome of surgical site infection diagnosed clinically (Mitchell 1986; Sekhar 2001; Halpern 2007; Lacasa 2007; Bezerra 2011; López-Cedrún 2011; Pasupathy 2011). Antibiotics were administered intravenously in one study (Halpern 2007) immediately prior to the procedure, and in the other six trials antibiotics were administered orally 1-2 hours prior to surgery. The pooled estimate showed a statistically significant reduction in infection in the antibiotic groups with risk ratio (RR) 0.29 (95% confidence interval (CI) 0.15 to 0.54) P = 0.0001.

#### **Post-operative antibiotics**

Four trials were included in this group (Sekhar 2001; Arteagoitia 2005; Lacasa 2007; López-Cedrún 2011). There were no infections in either the antibiotic or the placebo group in the trial by Sekhar, and the pooled estimate for the other three trials showed fewer infections in the antibiotic groups (RR 0.15 (95% CI 0.07 to 0.31) P < 0.00001).

#### Pre- and post-operative antibiotics

Two trials (Happonen 1990; Leon Arcila 2001) administered antibiotic or placebo both before and after the tooth extraction procedure and there was no difference between the infections reported in each group (RR 1.09 (95% CI 0.40 to 2.94) P = 0.87).

Overall the pooled estimate from all 10 trials which reported the outcome of infection showed that the use of antibiotics reduced the risk of infection by approximately 70% (RR 0.29 (95% CI 0.16 to 0.50) P < 0.0001). In individual trials the rate of infections in the placebo group varied between 0 and 56% with a mean event rate of 12.5% (Additional Table 3).

#### Dry socket

#### **Pre-operative antibiotics**

Six trials in this group reported this outcome and in two of these (Halpern 2007; López-Cedrún 2011) there were no dry sockets identified in either group. The pooled estimate for the other four trials (Ritzau 1992; Bergdahl 2004; Kaczmarzyk 2007; Bezerra 2011) showed no evidence of a difference between pre-operative antibiotics and placebo (RR 0.75 (95% CI 0.42 to 1.33) P = 0.32) with no statistical heterogeneity.

#### **Post-operative antibiotics**

Two trials administering antibiotics after the tooth extraction reported this outcome. The trial by López-Cedrún 2011 again reported no dry sockets in either group and the remaining trial (Arteagoitia 2005) showed no evidence of a difference (RR 0.18 (95% CI 0.01 to 3.70) P = 0.27).

#### Pre- and post-operative antibiotics

In three trials (Bystedt 1981; Barclay 1987; Kaczmarzyk 2007) antibiotics or placebo were administered prior to the tooth extraction and continued for 5 days post-operatively. The pooled estimate showed a reduction in the risk of dry socket (RR 0.52 (95% CI 0.27 to 0.99) P = 0.04) with no statistical heterogeneity.

The pooled estimate for all nine trials that reported the outcome of dry socket is RR 0.62 (95% CI 0.41 to 0.95) P = 0.03 with no statistical heterogeneity. This is a reduction in the risk of dry

socket from a mean of 6.9% in the placebo groups to 3.8% in the antibiotic groups (Additional Table 4).

# Pain (present or absent) days 6-7 or mean VAS score day 7

#### **Pre-operative antibiotics**

One trial in this outcome group reported pain as either present or absent and found no difference between the antibiotic and placebo groups (Analysis 1.3, Subgroup 1.3.1).

Likewise the three trials which reported mean pain score (visual analogue scale (VAS)) in each group at 7 days showed no difference between antibiotic and placebo mean difference (MD) -7.41 (95% CI -16.18 to 1.36) P = 0.10 (Analysis 1.4, Subgroup 1.4.1). There was moderate statistical heterogeneity in this analysis which may be due to the different study designs included or different antibiotics used in these trials.

#### **Post-operative antibiotics**

Two trials (Sekhar 2001; Arteagoitia 2005) in this group reported pain as a dichotomous outcome and also found no difference between antibiotic and placebo RR 0.51 (95% CI 0.14 to 1.82) P = 0.30 with substantial statistical heterogeneity ( $I^2 = 75\%$ ) (Analysis 1.3, Subgroup 1.3.2). This heterogeneity might be attributed to the different antibacterial spectrum of the two drugs. Arteagoitia 2005 used amoxicillin/clavulanic acid, which has a broad spectrum, and Sekhar 2001 used metronidazole which is active only against anaerobic bacteria.

#### Pre- and post-operative antibiotics

Only one study reported the dichotomous pain outcome in this group and found a benefit favouring the antibiotic group (Bystedt 1981) (Analysis 1.3, Subgroup 1.3.3).

In the four trials which reported the mean pain score at day 7 (Barclay 1987; Kaczmarzyk 2007; Bezerra 2011; López-Cedrún 2011), there was a reduction in pain in the antibiotic groups MD -8.17 (95% CI -11.90 to -4.45) P < 0.001 with no statistical heterogeneity (Analysis 1.4) (Additional Table 5).

#### Fever day 7

Results from four trials (Bystedt 1981; Happonen 1990; Arteagoitia 2005; Lacasa 2007) which included a combined total of 816 participants reported fever as a dichotomous outcome at day 7. The time of administration of antibiotics varied: pre-operative administration in Lacasa 2007, post-operative administration in Arteagoitia 2005 and both pre- and post-operative administration in Bystedt 1981 and Happonen 1990. In two of these trials there were no cases of fever in either group. The pooled estimate for the other two trials (Happonen 1990; Arteagoitia 2005) showed no evidence of a difference in post-operative fever between antibiotic and placebo groups RR 0.34 (95% CI 0.06 to 1.99) P = 0.23. Statistical heterogeneity was substantial ( $I^2 = 60\%$ ) and this is likely due to the different antibiotics and the varied times the antibiotics were administered.

#### Swelling day 7

#### **Pre-operative antibiotics**

Three trials (Sekhar 2001; Kaczmarzyk 2007; López-Cedrún 2011) including a total of 165 participants, comparing pre-operative antibiotics with placebo, found no evidence of a difference in swelling after 7 days between the antibiotic and placebo groups RR 1.13 (95% CI 0.69 to 1.83) P = 0.63 (Analysis 1.6, Subgroup 1.6.1).

#### **Post-operative antibiotics**

Only one trial in this group (Sekhar 2001) reported the outcome of swelling and found no difference between antibiotic and placebo groups (Analysis 1.6, Subgroup 1.6.2).

#### Pre- and post-operative antibiotics

There was no evidence of a difference in swelling after 7 days in the two trials (Kaczmarzyk 2007; López-Cedrún 2011) which reported this outcome RR 1.07 (95% CI 0.53 to 2.17) P = 0.85 (Analysis 1.6, Subgroup 1.6.3).

The pooled estimate for all groups in all three trials (combined total of 334 participants) showed no evidence of a difference between antibiotic and placebo groups for the outcome of swelling (RR 0.92 (95% CI 0.65 to 1.30) P = 0.63).

#### Trismus (dichotomous) day 7

The presence or absence of trismus was reported in only two trials (175 participants) in this review (Kaczmarzyk 2007; Pasupathy 2011) and found no evidence of a difference between antibiotic and placebo groups (RR 0.84 (95% CI 0.42 to 1.71) P = 0.64).

#### Adverse effects

Adverse effects were reported per participant by only 5 of the 18 trials included in this review. In a total of 930 participants there were twice as many people experiencing adverse effects in the antibiotic groups RR 1.98 (95% CI 1.10 to 3.59) P = 0.02. Reported adverse effects were generally mild and required no further treatment. However, in one trial (Kaczmarzyk 2007) 3% of participants taking a 5-day course of clindamycin developed gastric complications and were excluded from the trial.

# DISCUSSION

#### Summary of main results

This review included 18 double-blind placebo-controlled trials with a total of 2456 participants undergoing extraction of third molar (wisdom) teeth. None of the included studies were of patients undergoing tooth extraction in general dental practice, for the removal of severely decayed teeth. Thirteen of the included trials were at high risk of bias and the remaining five were at unclear risk of bias. There is evidence that antibiotics, administered to prevent infection in patients undergoing wisdom tooth extraction, reduce the risk of infection by approximately 70% (risk ratio (RR) 0.29 (95% confidence interval (CI) 0.16 to 0.50) P < 0.0001), and reduce the risk of dry socket by about one third (RR 0.62 (95% CI 0.41 to 0.95) P = 0.03). There is also evidence that patients who have antibiotics have overall less pain 7 days after the extraction compared to those receiving placebo, mean difference (MD) -8.17 (95% CI -11.90 to -4.45) which may be a direct result of the lower risk of infection (Summary of findings for the main comparison).

There is no evidence of a difference between antibiotics and placebo in the outcomes of fever (RR 0.34, 95% CI 0.06 to 1.99), swelling (RR 0.92, 95% CI 0.65 to 1.30) or trismus (RR 0.84, 95% CI 0.42 to 1.71) 7 days after tooth extraction. However, antibiotics are associated with an increase in generally mild and transient adverse effects compared to placebo (RR 1.98 (95% CI 1.10 to 3.59) P = 0.02).

While antibiotic prophylaxis is shown to reduce the risk of infection and dry socket, these outcomes still occur in some healthy people who take antibiotic prophylaxis associated with the extraction of impacted third molars. It is interesting to note that the rate of infection in the placebo groups in the included trials varied between nil (Leon Arcila 2001; Sekhar 2001) and 56% (Mitchell 1986) with a mean of 11.8% across the placebo groups of the included studies (Additional Table 3). Based on the evidence presented in this review the use of prophylactic antibiotics will reduce infection to a mean of 3%, which means that approximately 12 (range 10 to 17) people would need to receive antibiotic prophylaxis to prevent one infection.

The incidence of dry socket in the placebo group varied between nil (Halpern 2007; López-Cedrún 2011) and 34% (Barclay 1987) with a mean of 6.9%. This means that approximately 38 (range 24 to 250) healthy people would need to be treated with prophylactic antibiotics to prevent one case of dry socket (Additional Table 4). However using prophylactic antibiotics is likely to result in at least one adverse effect for every 21 people treated (range 8 to 200), though adverse effects reported in the trials were generally mild and transient.

# Overall completeness and applicability of evidence

We conducted a comprehensive search including both electronic and handsearching through reference lists. We identified 18 randomised double-blind placebo-controlled trials including a combined total of approximately 2500 participants. Trials were conducted in different countries but included healthy patients in their 20s, undergoing extraction of impacted teeth (mainly of the lower jaw), thus making the results of our review quite sound regarding effectiveness of antibiotic prophylaxis of infectious complications in healthy young people undergoing wisdom tooth extractions, actually a very large proportion of surgical tooth extractions.

However, we identified no trials of patients attending general dental practices for the extraction of teeth due to caries or periodontitis. Identified trials did not include patients with depressed immune systems, patients with other illnesses, young children or elderly patients who required tooth extractions. Indeed, it is unlikely to be feasible or ethical to conduct placebo-controlled trials in this group of patients. The results of this review may or may not be generalisable to this group who would be expected to be at higher risk of infection. However, on the basis of the results of this review, it is likely that in subjects at higher risk of infectious complications, antibiotic prophylaxis may be more effective and the number of people needed to receive antibiotics to prevent one infection likely to decrease.

Another limit to generalisability of our results regards the clinical skill of the operators, as those in the included studies were mainly oral surgery specialists working in referral centres. Whether results would be similar for general dental practitioners is unclear.

Adverse effect frequency and severity can be important determinants in deciding about a preventive treatment. As for many medical areas, quality and quantity of information about adverse effects of interventions in these trials were inadequate (Ioannidis 2009). However, on the basis of the drop-out rates, and the adverse effects in the five trials which reported adverse effects per patient, it seems likely that adverse effects were generally mild and well tolerated. This review cannot provide any information on the extent to which the use of prophylactic antibiotics in association with tooth extraction in healthy people may have on the subsequent development of strains of bacteria resistant to antibiotics in common use in these situations (EU Commission 2011).

# Quality of the evidence

Although this review was restricted to double-blind placebo-controlled trials none of the included trials were at low risk of bias (unclear (5 trials) or high (13 trials)). The most common sources of bias were missing outcome data and selective reporting. In trials such as many of those included in this review, where the outcome events are uncommon even in the placebo group, losses to followup can potentially cause misleading results.

The quality of the body of evidence included in this review was evaluated using the GRADE system in Summary of findings for the main comparison. For most outcomes the quality of the evidence was moderate, because of high or unclear risk of bias in the studies. For the outcomes of fever and presence of pain on day 6-7, there was also heterogeneity between studies, so the quality of the body of evidence for these outcomes was graded as low.

We found no evidence concerning the use of prophylactic antibiotics in patients undergoing extraction for severe caries or periodontitis.

#### Potential biases in the review process

Data from some of the studies included in the present review, namely older ones, could not be entered in the meta-analysis because of the poor reporting that prevented data extraction. This may have introduced a reporting bias into this review. The funnel plot for the primary outcome of infection (Figure 4) shows no evidence of publication bias (note the points on the plot are not independent because two of the trials (Lacasa 2007; López-Cedrún 2011) are included in two subgroups).

Figure 4. Funnel plot of comparison: I Antibiotic versus placebo, outcome: I.I Local sign of infection.



There were some post hoc changes from the protocol for this review, and we acknowledge that such changes can potentially introduce a bias into the review process (*see* Differences between protocol and review). The inclusion criteria for the review were amended so that only randomised, double-blind placebo-controlled trials were included. The protocol planned to only include trials where the important clinical outcome of infection was reported. In this review we made it more explicit that trials which only reported other or intermediate outcomes (endocarditis incidence, bacteraemia or serum marker of infection) would be excluded. We think that these changes have resulted in higher quality clinically relevant trials being included in our review.

# Agreements and disagreements with other studies or reviews

A systematic review on the same subject was published in 2007 (Ren 2007). This review included a different group of studies, because of different inclusion criteria (they considered mandibular third molar extractions only and did not limit the review to double-blind studies). Ren 2007 concluded that antibiotic administration was effective in preventing wound infection, although they reported a higher number needed to treat: "on average 25 patients needed to be treated with systemic antibiotics to prevent 1 case of extraction wound infection" in this group of healthy patients.

# AUTHORS' CONCLUSIONS

#### Implications for practice

There is moderate quality evidence that the use of prophylactic antibiotics reduces the risk of infectious complications following third molar extraction. There is no clear evidence that timing of antibiotic administration (pre-operative, post-operative or both) is important. The numbers of healthy people undergoing third molar extraction who need to be treated with antibiotics to prevent one infection range between 10 and 17, and to prevent a case of dry socket between 24 and 250 people would need to receive prophylactic antibiotics. The size of the benefit is not enough to recommend a routine use of this practice, due to the increased risk of mild adverse effects for the patients and also the potential for contributing to the development of bacterial resistance.

#### Implications for research

Future trials should investigate prophylactic antibiotics effectiveness in patients at high risk of infective complications, such as immunocompromised subjects and patients who have experienced infective complications following previous extractions. Trials on patients undergoing extractions for severe caries or periodontal

disease are also needed. Future studies should also measure the outcomes of symptoms and clinical assessment using standardised measures and timepoints, and report these according to CON-SORT guidelines.

# ACKNOWLEDGEMENTS

Our thanks to Luisa Fernandez-Mauleffinch who translated two papers from Spanish to English, and to Annette Blumle who translated a paper from German to English. Our thanks to Jo Leese for her contributions to the plain language summary of this review.

#### REFERENCES

#### References to studies included in this review

#### Arteagoitia 2005 {published data only}

Arteagoitia I, Diez A, Barbier L, Santamaria G, Santamaria J. Efficacy of amoxicillin/clavulanic acid in preventing infectious and inflammatory complications following impacted mandibular third molar extraction. *Oral Surgery, Oral Medicine, Oral Pathology, Oral Radiology, and Endodontology* 2005;**100**(1):e11–8.

#### Barclay 1987 {published data only}

Barclay JK. Metronidazole and dry socket: prophylactic use in mandibular third molar removal complicated by nonacute pericoronitis. *New Zealand Dental Journal* 1987;**83**: 71–5.

#### Bergdahl 2004 {published data only}

Bergdahl M, Hedstrom L. Metronidazole for the prevention of dry socket after removal of partially impacted mandibular third molar: a randomised controlled trial. *British Journal* of Oral & Maxillofacial Surgery 2004;**42**(6):555–8.

#### Bezerra 2011 {published and unpublished data}

Bezerra T. Poster Board Number: 16: Prophylaxis versus placebo treatment for infective and inflammatory complications of surgical third molar removal: A splitmouth, double-blind, controlled, clinical trial with amoxicillin (500 mg). *Journal of Oral and Maxillofacial Surgery* 2010;**68**(9 (suppl)):e69.

\* Bezerra TP, Studart-Soares EC, Scaparo HC, Pita-Neto IC, Batista SH, Fonteles CS. Prophylaxis versus placebo treatment for infective and inflammatory complications of surgical third molar removal: a split-mouth, double-blind, controlled, clinical trial with amoxicillin (500 mg). *Journal* of Oral & Maxillofacial Surgery 2011;69(11):e333–9.

#### Bystedt 1980 {published data only}

Bystedt H, Nord CE, Nordenram A. Effect of azidocillin, erythromycin, clindamycin and doxycycline on postoperative complications after surgical removal of impacted mandibular third molars. International Journal of Oral Surgery 1980; Vol. 9, issue 3:157–65.

#### Bystedt 1981 {published data only}

Bystedt H, von Konow L, Nord CE. A comparison of the effect of phenoxymethylpenicillin and azidocillin on postoperative complications after surgical removal of impacted mandibular third molars. *Swedish Dental Journal* 1981;**5**(5-6):225–34.

#### Halpern 2007 {published data only}

Halpern LR, Dodson TB. Does prophylactic administration of systemic antibiotics prevent postoperative inflammatory complications after third molar surgery?. *Journal of Oral and Maxillofacial Surgery* 2007;**65**(2):177–85.

#### Happonen 1990 {published data only}

Happonen RP, Backstrom AC, Ylipaavalniemi P. Prophylactic use of phenoxymethylpenicillin and tinidazole in mandibular third molar surgery, a comparative placebo controlled clinical trial. *British Journal of Oral & Maxillofacial Surgery* 1990;**28**(1):12–5.

#### Kaczmarzyk 2007 {published data only}

Kaczmarzyk T, Wichlinski J, Stypułkowska J, Zaleska M, Panas M, Woron J. Single-dose and multi-dose clindamycin therapy fails to demonstrate efficacy in preventing infectious and inflammatory complications in third molar surgery. *International Journal of Oral and Maxillofacial Surgery* 2007; **36**(5):417–22.

#### Kaziro 1984 {published data only}

\* Kaziro GS. Metronidazole (Flagyl) and Arnica Montana in the prevention of post-surgical complications, a comparative placebo controlled clinical trial. *British Journal of Oral & Maxillofacial Surgery* 1984;**22**(1):42–9.

Kaziro GS, Harris M. A comparative placebo controlled trial between metronidazole and Arnica Montana in preventing post-surgical complications. *Journal of Dental Research* 1981;**60**(Special Issue):1176.

#### Lacasa 2007 {published data only}

Lacasa JM, Jiménez JA, Ferrás V, Bossom M, Sóla-Morales O, García-Rey C, et al.Prophylaxis versus pre-emptive treatment for infective and inflammatory complications of surgical third molar removal: a randomized, double-blind,

placebo-controlled, clinical trial with sustained release amoxicillin/clavulanic acid (1000/62.5 mg). *International Journal of Oral and Maxillofacial Surgery* 2007;**36**(4):321–7.

#### Leon Arcila 2001 {published and unpublished data}

Leon Arcila ME, Acosta P, Bravo BE, Mena A, Noguera DP, Zúñiga JR. Antibiotic prophylaxis in third molar surgery [Profilaxis antibiotica en cirugia de terceros molares incluidos]. Revista Estomatologia 2001; Vol. 9, issue 2: 4–13.

#### López-Cedrún 2011 {published and unpublished data}

López-Cedrún JL, Pijoan JI, Fernández S, Santamaria J, Hernandez G. Efficacy of amoxicillin treatment in preventing postoperative complications in patients undergoing third molar surgery: a prospective, randomized, double-blind controlled study. *Journal of Oral and Maxillofacial Surgery* 2011;**69**:e5–e14.

### MacGregor 1980 {published data only}

\* Macgregor AJ, Addy A. Value of penicillin in the prevention of pain, swelling and trismus following the removal of ectopic mandibular third molars. *International Journal of Oral Surgery* 1980;**9**(3):166–72. MacGregor AJ, Addy AG. The effect of penicillin upon pain, swelling and trismus following the removal of impacted third molars. *Journal of Dental Research* 1980;**59** (Special Issue D Part 1):1799 (Abs No 35).

#### Mitchell 1986 {published data only}

Mitchell DA. A controlled clinical trial of prophylactic tinidazole for chemoprophylaxis in third molar surgery. *British Dental Journal* 1986;**160**(8):284–6.

#### Pasupathy 2011 {published data only}

Pasupathy S, Alexander M. Antibiotic prophylaxis in third molar surgery. *Journal of Craniofacial Surgery* 2011;**22**: 551–3.

#### Ritzau 1992 {published data only}

Ritzau M, Hillerup S, Branebjerg PE, Ersbol BK. Does metronidazole prevent alveolitis sicca dolorosa? A doubleblind, placebo-controlled clinical study. *International Journal of Oral and Maxillofacial Surgery* 1992;**21**(5): 299–302.

#### Sekhar 2001 {published data only}

Sekhar CH, Narayanan V, Baig MF. Role of antimicrobials in third molar surgery: prospective, double blind, randomized, placebo-controlled clinical study. British Journal of Oral & Maxillofacial Surgery 2001; Vol. 39, issue 2:134–7.

#### References to studies excluded from this review

#### Abu-Mowais 1990 {published data only}

Abu-Mowais M, Abdul-Amir R, El-Mostehy MR, Mansour M, Mohalhal I. The clinical safety and efficacy of antibiotic prophylaxis after surgical removal of third molars. Journal of the Kuwait Medical Association 1990; Vol. 24, issue 1: 57–9.

#### Ataoglu 2008 {published data only}

Ataoğ lu H, Oz GY, Candirli C, Kiziloğ lu D. Routine antibiotic prophylaxis is not necessary during operations to remove third molars. British Journal of Oral & Maxillofacial Surgery 2008;46(2):133–5.

#### Bargnesi 1985 {published data only}

Bargnesi R, Bondi C, Cavalli A, Vallania G. A clinical double blind study of the usefulness of systemic prophylactic antibiotic therapy in minor oral surgery. *Giornale di Stomatologia e di Ortognatodonzia* 1985;4(2):29–32.

#### Curran 1974 {published data only}

Curran JB, Kennett S, Young AR. An assessment of the use of prophylactic antibiotics in third molar surgery. International Journal of Oral Surgery 1974; Vol. 3, issue 1: 1–6.

#### Delilbasi 2004 {published data only}

Delilbasi C, Saracoglu U, Keskin A. Effects of 0.2% chlorhexidine gluconate and amoxicillin plus clavulanic acid on the prevention of alveolar osteitis following mandibular third molar extractions. *Oral Surgery, Oral Medicine, Oral Pathology, Oral Radiology, & Endodontology* 2002;**94**(3): 301–4.

#### de Moura 2011 {published data only}

de Moura WL, de Moura WL, Freire SASR, Mendes SM, Olate S. Efficacy of amoxicillin treatment in preventing postoperative complications in patients undergoing third molar surgery: a double blind study [Eficacia del tratamiento con amoxicilina en la prevención de complicaciones postoperatorias en pacientes sometidos a cirugía del tercer molar: un estudio doble ciego]. *International Journal of Odontostomatology* 2011;5(2):147–52.

#### Foy 2004 {published data only}

Foy SP, Shugars DA, Phillips C, Marciani RD, Conrad SM, White RP Jr. The impact of intravenous antibiotics on health-related quality of life outcomes and clinical recovery after third molar surgery. *Journal of Oral and Maxillofacial Surgery* 2004;**62**(1):15–21.

#### Fridrich 1990 {published data only}

Fridrich KL, Olson RA. Alveolar osteitis following surgical removal of mandibular third molars. *Anesthesia Progress* 1990;**37**(1):32–41.

#### Graziani 2005 {published data only}

Graziani F, Corsi L, Fornai M, Antonioli L, Tonelli M, Ceia S, et al.Clinical evaluation of piroxicam-FDDF and azithromycin in the prevention of complications associated with impacted lower third molar extraction. *Pharmacological Research* 2005;**52**(6):485–90.

#### Grossi 2007 {published data only}

Grossi GB, Maiorana C, Garramone RA, Borgonovo A, Creminelli L, Santoro F. Assessing postoperative discomfort after third molar surgery: a prospective study. *Journal of Oral and Maxillofacial Surgery* 2007;**65**(5):901–17.

#### Head 1984 {published data only}

Head TW, Bentley KC, Millar EP, deVries JA. A comparative study of the effectiveness of metronidazole and penicillin V in eliminating anaerobes from postextraction bacteremias. *Oral Surgery, Oral Medicine, and Oral Pathology* 1984;**58**(2): 152–5.

#### Krekmanov 1980 {published data only}

Krekmanov L, Hallander HO. Relationship between bacterial contamination and alveolitis after third molar surgery. *International Journal of Oral Surgery* 1980;**9**(4): 274–80.

#### Krekmanov 1981 {published data only}

Krekmanov L. Alveolitis after operative removal of third molars in the mandible. *International Journal of Oral Surgery* 1981;**10**(3):173–9.

#### Krekmanov 1986 {published data only}

Krekmanov L, Nordenram A. Postoperative complications after surgical removal of mandibular third molars. Effects of penicillin V and chlorhexidine. *International Journal of Oral and Maxillofacial Surgery* 1986;**15**(1):25–9.

#### Laird 1972 {published data only}

Laird WR, Stenhouse D, Macfarlane TW. Control of post-operative infection. A comparative evaluation of clindamycin and phenoxymethylpenicillin. British Dental Journal 1972; Vol. 133, issue 3:106–9.

#### Limeres 2009 {published data only}

Limeres J, Sanromán JF, Tomás I, Diz P. Patients' perception of recovery after third molar surgery following postoperative treatment with moxifloxacin versus amoxicillin and clavulanic acid: a randomized, double-blind, controlled study. *Journal of Oral and Maxillofacial Surgery* 2009;**67**(2): 286–91.

#### Lombardia Garcia 1987 {published data only}

Lombardía García E, García Pola MJ, González García M, Gener González M. Antimicrobial prophylaxis in surgery of the third molar. Analytic study of postoperative complications. *Archivos de Odonto Estomatologia* 1987;**3**(3): 130–4.

#### Lopes 2011 {published data only}

Lopes DR, Peres MP, Levin AS. Randomized study of surgical prophylaxis in immunocompromised hosts. *Journal of Dental Research* 2011;**90**(2):225–9.

#### Luaces-Rey 2010 {published data only}

Luaces-Rey R, Arenaz-Búa J, Lopez-Cedrun-Cembranos JL, Martínez-Roca C, Pértega-Díaz S, Sironvalle-Soliva S. Efficacy and safety comparison of two amoxicillin administration schedules after third molar removal. A randomized, double-blind and controlled clinical trial. *Medicina Oral, Patologia Oral y Cirugia Bucal* 2010;**15**(4): 633–8.

#### Lyall 1991 {published data only}

Lyall JB. Third molar surgery: the effect of primary closure, wound dressing and metronidazole on postoperative recovery. *Journal of the Royal Army Medical Corps* 1991;**137** (2):100–3.

#### MacGregor 1973 {published data only}

MacGregor AJ, Hutchinson D. The effect of Nivemycin on pain and swelling following lower third molar removal. British Journal of Oral Surgery 1973; Vol. 10, issue 3: 321–5.

#### Mitchell 1987 {published data only}

Mitchell DA, Morris TA. Tinidazole or pivampicillin in third molar surgery. *International Journal of Oral and Maxillofacial Surgery* 1987;**16**(2):171–4.

#### Monaco 1999 {published data only}

Monaco G, Staffolani C, Gatto MR, Checchi L. Antibiotic therapy in impacted third molar surgery. *European Journal of Oral Science* 1999;**6**(107):437–41.

#### Monaco 2009 {published data only}

Monaco G, Tavernese L, Agostini R, Marchetti C. Evaluation of antibiotic prophylaxis in reducing postoperative infection after mandibular third molar extraction in young patients. *Journal of Oral and Maxillofacial Surgery* 2009;**67**(7):1467–72.

#### Olusanya 2011 {published data only}

Olusanya AA, Arotiba JT, Fasola OA, Akadiri AO. Prophylaxis versus pre-emptive antibiotics in third molar surgery: a randomised control study. Nigerian Postgraduate Medical Journal 2011; Vol. 18, issue 2:105–10.

#### Osborn 1979 {published data only}

Osborn JF, Jaques WA, Thalmann R, Spiessl B. Efficiency of systemic prophylactic antibiotics in the surgical removal of lower wisdom teeth. *SSO: Schweizerische Monatsschrift fur Zahnheilkunde* 1979;**89**(5):458–66.

#### Poeschl 2004 {published data only}

Poeschel PW, Eckel D. Antibiotic prophylaxis after outpatient removal of wisdom teeth: a necessity?. *Stomatologie* 2003;**100**(5):115.

\* Poeschl PW, Eckel D, Poeschl E. Postoperative prophylactic antibiotic treatment in third molar surgery--a necessity?. *Journal of Oral and Maxillofacial Surgery* 2004; **62**(1):3–9.

#### Reekie 2006 {published data only}

Reekie D, Downes P, Devlin CV, Nixon GM, Devlin H. The prevention of 'dry socket' with topical metronidazole in general dental practice. *British Dental Journal* 2006;**200**(4): 210–3.

#### Rood 1979 {published data only}

Rood JP, Murgatroyd J. Metronidazole in the prevention of 'dry socket'. British Journal of Oral Surgery 1979; Vol. 17, issue 1:62–70.

#### Samsudin 1994 {published data only}

Samsudin AR, Mason DA. Symptoms from impacted wisdom teeth. *British Journal of Oral & Maxillofacial Surgery* 1994;**32**(6):380–3.

#### Siddiqi 2010 {published data only}

Siddiqi A, Morkel JA, Zafar S. Antibiotic prophylaxis in third molar surgery: A randomized double-blind placebocontrolled clinical trial using split-mouth technique. *International Journal of Oral and Maxillofacial Surgery* 2010; **39**(2):107–4.

#### Stavropoulos 2006 {published data only}

Stavropoulos MF, Shugars DA, Phillips C, Conrad SM, Fleuchaus PT, White RP Jr. Impact of topical minocycline with third molar surgery on clinical recovery and health-

related quality of life outcomes. *Journal of Oral and Maxillofacial Surgery* 2006;**64**(7):1059–65.

#### Sulejmanagic 2005 {published data only}

Sulejmanagić N, Sulejmanagić H, Ljutović Z, Salihagi

c D, Sijercić M. Combined application of amoxicillin and clavulanic acid after oral surgical interventions. *Bosnian Journal of Basic Medical Science* 2005;5(1):61–8.

#### Swanson 1989 {published data only}

Swanson AE. A double-blind study on the effectiveness of tetracycline in reducing the incidence of fibrinolytic alveolitis. *Journal of Oral and Maxillofacial Surgery* 1989;47 (2):165–7.

#### Uluibau 2005 {published data only}

Uluibau IC, Jaunay T, Goss AN. Severe odontogenic infections. *Australian Dental Journal* 2005;**50**(2):S74–S81.

#### Walkow 1995 {published data only}

Walkow T, Papageorge M, Norris L, Cottrell D. Prophylactic antibiotic efficacy in third molar surgery (AADR Abstract 1995). *Journal of Dental Research* 1995;74(Special Issue): 138 (Abs No 1013).

#### Yoshii 2002 {published data only}

Yoshii T, Hamamoto Y, Muraoka S, Furudoi S, Komori T. Differences in postoperative morbidity rates, including infection and dry socket, and differences in the healing process after mandibular third molar surgery in patients receiving 1-day or 3-day prophylaxis with lenampicillin. Journal of Infection & Chemotherapy 2002; Vol. 8, issue 1: 87–93.

#### References to ongoing studies

#### Arteagoitia 2011 {published data only}

Arteagoitia M, Santamaria G, Ramos E, Barbier L, Alvarez J, Martin J, et al.Efficacy of amoxicillin/clavulanic acid 2000/125 mg in preventing infection after extraction of impacted mandibular third molar totally covered by bone: preliminary results. *International Journal of Oral and Maxillofacial Surgery* 2011;**40**(10):1048–9.

#### Additional references

#### Aida 2009

Aida J, Morita M, Akhter R, Aoyama H, Masui M, Ando Y. Relationships between patient characteristics and reasons for tooth extraction in Japan. *Community Dental Health* 2009;**26**(2):104–9.

#### Akhter 2008

Akhter R, Hassan NM, Aida J, Zaman KU, Morita M. Risk indicators for tooth loss due to caries and periodontal disease in recipients of free dental treatment in an adult population in Bangladesh. *Oral Health & Preventive Dentistry* 2008;**6** (3):199–207.

#### Al-Shammari 2006

Al-Shammari KF, Al-Ansari JM, Al-Melh MA, Al-Khabbaz AK. Reasons for tooth extraction in Kuwait. *Medical Principles & Practice* 2006;15(6):417–22.

#### Anand 2010

Anand PS, Kamath KP, Nair B. Trends in extraction of permanent teeth in private dental practices in Kerala state, India. *Journal of Contemporary Dental Practice* 2010;**11**(3): 41–8.

#### Baqain 2007

Baqain ZH, Khraisat A, Sawair F, Ghanam S, Shaini FJ, Rajab LD. Dental extraction for patients presenting at oral surgery student clinic. *Compendium of Continuing Education in Dentistry* 2007;**28**(3):146–50.

#### Bortoluzzi 2010

Bortoluzzi MC, Manfro R, De Déa BE, Dutra TC. Incidence of dry socket, alveolar infection, and postoperative pain following the extraction of erupted teeth. *Journal of Contemporary Dental Practice* 2010;**11**(1):E033–40.

#### Bouloux 2007

Bouloux GF, Steed MB, Perciaccante VJ. Complications of third molar surgery. *Oral and Maxillofacial Surgery Clinics of North America* 2007;**19**(1):117–28.

#### Byahatti 2011

Byahatti SM, Ingafou MS. Reasons for extraction in a group of Libyan patients. *International Dental Journal* 2011;**61** (4):199–203.

#### Chestnutt 2000

Chestnutt IG, Binnie VI, Taylor MM. Reasons for tooth extraction in Scotland. *Journal of Dentistry* 2000;**28**(4): 295–7.

#### Chrysanthakopoulos 2011

Chrysanthakopoulos NA. Reasons for extraction of permanent teeth in Greece: a five-year follow-up study. *International Dental Journal* 2011;**61**(1):19–24.

#### Costelloe 2010

Costelloe C, Metcalfe C, Lovering A, Mant D, Hay AD. Effect of antibiotic prescribing in primary care on antimicrobial resistance in individual patients: systematic review and meta-analysis. *BMJ* 2010;**340**:c2096.

#### Da'ameh 2006

Da'ameh D. Reasons for permanent tooth extraction in the north of Afghanistan. *Journal of Dentistry* 2006;**34**(1): 48–51.

#### Daly 2008

Daly B, Newton T, Nasser M, Jones K, Fedorowicz Z. Intrasocket interventions for the treatment of dry socket. *Cochrane Database of Systematic Reviews* 2008, Issue 1. [DOI: 10.1002/14651858.CD006968]

#### Danielson 2011

Danielson OE, Chinedu AC, Oluyemisi EA, Bashiru BO, Ndubuisi OO. Frequency, causes and pattern of adult tooth extraction in a Nigerian rural health facility. *Odontostomatologie Tropicale* 2011;**34**(134):5–10.

#### Dar-Odeh 2010

Dar-Odeh NS, Abu-Hammad OA, Al-Omiri MK, Khraisat AS, Shehabi AA. Antibiotic prescribing practices by dentists: a review. *Therapeutics and Clinical Risk Management* 2010; **6**:301–6.

#### Egger 1997

Egger M, Davey Smith G, Schneider M, Minder C. Bias in meta-analysis detected by a simple, graphical test. *BMJ* 1997;**315**(7109):629–34.

#### Elbourne 2002

Elbourne DR, Altman DG, Higgins JPT, Curtin F, Worthington HV, Vaillancourt JM. Meta-analyses involving cross-over trials: methodological issues. *International Journal of Epidemiology* 2002;**31**:140–9.

#### Epstein 2000

Epstein JB, Chong S, Le ND. A survey of antibiotic use in dentistry. *Journal of the American Dental Association* 2000; **131**(11):1600–9.

#### EU Commission 2011

European Commission. Directorate-General for Health & Consumers. Communication from the European Commission to the European Parliament and the Council. Action plan against the rising threats from Antimicrobial Resistance (COM (2011) 748). http://ec.europa.eu/dgs/health<sup>\*</sup>consumer/docs/ communication<sup>\*</sup>amr<sup>\*</sup>2011<sup>\*</sup>748<sup>\*</sup>en.pdf (accessed July 2012).

# Hersh 1999

Hersh EV. Adverse drug interactions in dental practice: interactions involving antibiotics. *Journal of the American Dental Association* 1999;**130**(2):236–51.

#### Higgins 2011

Higgins JPT, Green S (editors). Cochrane Handbook for Systematic Reviews of Interventions version 5.1.0 (updated March 2011). The Cochrane Collaboration, 2011. Available from www.cochrane-handbook.org.

#### Howe 1985

Howe LH. *Minor Oral Surgery*. 3rd Edition. Bristol: Wright, 1985.

#### Ioannidis 2009

Ioannidis JP. Adverse events in randomized trials: neglected, restricted, distorted, and silenced. *Archives of Internal Medicine* 2009;**169**(19):1737–9.

#### Jaafar 2000

Jaafar N, Nor GM. The prevalence of post-extraction complications in an outpatient dental clinic in Kuala Lumpur Malaysia-a retrospective survey. *Singapore Dental Journal* 2000;**23**(1):24–8.

#### Jovino-Silveira 2005

Jovino-Silveira RC, Caldas Ade F Jr, de Souza EH, Gusmão ES. Primary reason for tooth extraction in a Brazilian adult population. *Oral Health & Preventive Dentistry* 2005;**3**(3): 151–7.

#### Karki 2011

Karki AJ, Holyfield G, Thomas D. Dental prescribing in Wales and associated public health issues. *British Dental Journal* 2011;**210**(1):E21.

#### Lesolang 2009

Lesolang RR, Motloba DP, Lalloo R. Patterns and reasons for tooth extraction at the Winterveldt Clinic: 1998-2002. *South African Dental Journal* 2009;**64**(5):214-5, 218.

#### Lökken 1975

Lökken P, Olsen I, Bruaset I, Norman-Pedersen K. Bilateral surgical removal of impacted lower third molar teeth as a model for drug evaluation: a test with ibuprofen. *European Journal of Clinical Pharmacology* 1975;**8**(3-4):209–16.

#### McCaul 2001

McCaul LK, Jenkins WMM, Kay EJ. The reasons for extraction of permanent teeth in Scotland: a 15-year followup study. *British Dental Journal* 2001;**190**(12):658–62.

#### Mettes 2012

Mettes TD, Ghaeminia H, Nienhuijs ME, Perry J, van der Sanden WJ, Plasschaert A. Surgical removal versus retention for the management of asymptomatic impacted wisdom teeth. *Cochrane Database of Systematic Reviews* 2012, Issue 6. [DOI: 10.1002/14651858.CD003879]

#### Ren 2007

Ren YF, Malmstrom HS. Effectiveness of antibiotic prophylaxis in third molar surgery: a meta-analysis of randomized controlled clinical trials. *Journal of Oral and Maxillofacial Surgery* 2007;**65**(10):1909–21.

#### Richards 2005

Richards W, Ameen J, Coll AM, Higgs G. Reasons for tooth extraction in four general dental practices in South Wales. *British Dental Journal* 2005;**198**(5):275–8.

#### Rücker 2008

Rucker G, Schwarzer G, Carpenter J. Arcsine test for publication bias in meta-analyses with binary outcomes. *Statistics in Medicine* 2008;**27**(5):746–63.

#### Sleeman 1995

Sleeman D, Barry H, Ryan C, Allen F. Day case oral surgery: a review of 25 years experience at the Dublin Dental Hospital. *British Dental Journal* 1995;**179**(6):221–4.

#### Thomas 1994

Thomas D, Walker R, Smith A, Shepherd J. The provision of oral surgery services in England and Wales 1984-1991. British Dental Journal 1994; Vol. 176, issue 6:215–9.

#### Trovik 2000

Trovik TA, Klock KS, Haugejorden O. Trends in reasons for tooth extractions in Norway from 1968 to 1998. *Acta Odontolologica Scandinavica* 2000;**58**(2):89–96.

#### Worthington 1999

Worthington H, Clarkson J, Davies R. Extraction of teeth over 5 years in regularly attending adults. *Community Dentistry and Oral Epidemiology* 1999;**27**(3):187–94.

\* Indicates the major publication for the study

# CHARACTERISTICS OF STUDIES

# **Characteristics of included studies** [ordered by year of study]

# MacGregor 1980

Methods	Study design: RCT where participants paired based on number of lower molars extracted Conducted in: Leeds, England. Number of centres: 1. Recruitment period: not stated.
Participants	Inclusion criteria: Caucasian patients requiring removal of 1 or 2 mandibular third molars under endotracheal anaesthesia. M3 had to be fully developed with an identifiable occlusal plane Exclusion criteria: patients who wear artificial dentures, who could not attend 4th day appointment or those whose operation had "undue haemorrhage". Patients who required antibiotics for other reasons (e.g. endocarditis) were also excluded Age group: not stated. Number randomised: not stated. Number evaluated: not stated.
Interventions	<b>Comparison: pre-op penicillin versus placebo.</b> Group A: benzyl penicillin 300 mg + procaine penicillin 300 mg intramuscular 30 min pre-operatively Group B: placebo injection IM 30 min pre-operatively. Procedures performed by 2 surgeons with attempts to standardise methods
Outcomes	Pain, swelling and trismus on day 4 in graphs only.
Notes	

# Risk of bias

Bias	Authors' judgement	Support for judgement
Random sequence generation (selection bias)	Unclear risk	Method of sequence generation not de- scribed.
Allocation concealment (selection bias)	Unclear risk	Not mentioned.
Blinding of participants and personnel (performance bias) All outcomes	Low risk	Double blind (both participants and sur- geons).
Blinding of outcome assessment (detection bias) All outcomes	Low risk	Double blind (both participants and sur- geons).
Incomplete outcome data (attrition bias) All outcomes	Unclear risk	Numbers of patients allocated to treatment and assessed on day 4 not stated

Antibiotics to prevent complications following tooth extractions (Review)

Copyright © 2013 The Cochrane Collaboration. Published by John Wiley & Sons, Ltd.

# MacGregor 1980 (Continued)

Selective reporting (reporting bias)	High risk	Planned report of pain swelling and trismus presented only in graphs. No estimates per group given		
Other bias	Low risk	No other sources of bias identified.		
Bystedt 1980				
Methods	Study design: RCT. Conducted in: Sweden. Number of centres: 1. Recruitment period: not stated.			
Participants	Inclusion criteria: healthy outpatients requiring surgical removal of impacted third molar of mandible Exclusion criteria: history of significant gastric, hepatic or renal disease, those taking any other medication except analgesia during study period Age group: mean 29 years, range 17-79 years. Number randomised: 140 in 3 separate subtrials. Number evaluated: unclear, reported as percentage of combined groups			
Interventions	Comparison A: 1 hour pre-op + 7 days post-op azidocillin versus placebo. Comparison B: 90 min pre-op + 7 days post-op erythromycin or clindamycin versus placebo. Comparison C: 180 min pre-op + 7 days post-op doxycycline versus placebo. Study A (n = 40): either azidocillin 750 mg 1 hour pre-op + 750 mg bid for 7 days post-op or matching placebo Study B (n = 60): either erythromycin stearate 500 mg or clindamycin 300 mg or placebo 90 min pre-op followed by 250 mg erythromycin or 150 mg clindamycin or placebo 4x daily for 7 days Study C (n = 40): either 200 mg doxycycline or placebo 180 min pre-op plus either 100 mg doxycycline or placebo once daily for 7 days All participants had 0.5-1 g acetylsalicylic acid as needed for pain			
Outcomes	Capillary serum antibiotic levels, dental alveolar blood antibiotic levels, bone antibiotic levels, evaluated on day 2. Duration of operation, pain, trismus, swelling, wound healing, side effects evaluated on days 2, 5 and 7 post-op			
Notes				
Risk of bias				
Bias	Authors' judgement	Support for judgement		
Random sequence generation (selection bias)	Unclear risk	Quote: "assigned at random". Method of sequence generation not described		

# Bystedt 1980 (Continued)

Allocation concealment (selection bias)	Unclear risk	Not mentioned.	
Blinding of participants and personnel (performance bias) All outcomes	Low risk	Quote: "Double blind".	
Blinding of outcome assessment (detection bias) All outcomes	Low risk	Quote: "Double blind".	
Incomplete outcome data (attrition bias) All outcomes	Unclear risk	Numbers of patients allocated to antibiotic or placebo not explicitly stated for each of the subtrials and numbers evaluated not stated for each subtrial for each outcome	
Selective reporting (reporting bias)	High risk	All planned outcomes reported, but not for each randomised treatment group, and no estimates of variance given for pain	
Other bias	Unclear risk	No description of characteristics of patients by randomised group at baseline	
Bystedt 1981			
Methods	Study design: 3-arm RCT. Conducted in: Sweden. Number of centres: 1. Recruitment period: unclear. Funding source: unspecified.		
Participants	Inclusion criteria: healthy outpatients, referred for surgical removal of an impacted third molar of the mandible Exclusion criteria: not specified. Age group: range 17-30 years. Group A: randomised 20; analysed 20. Group B: randomised 20; analysed 20. Group C: randomised 20; analysed 20.		
Interventions	Comparison: pre- + post-op penicillin versus pre- + post-op azidocillin versus placebo. Group A: phenoxmethylpenicillin 800 mg 1 hour before operation and then twice a day (at 9.00 AM and 9.00 PM) for 7 days Group B: azidocillin 750 mg 1 hour before operation and then twice a day (at 9.00 AM and 9.00 PM) for 7 days Group C: placebo 1 hour before operation and then twice a day (at 9.00 AM and 9.00 PM) for 7 days Aspirin 0.5-1.0 g was provided to all participants as a rescue analgesic to be taken when needed. No other medications except analgesics were allowed during the investigation		

# Bystedt 1981 (Continued)

	period
Outcomes	Pain was measured on the day of operation and on days 2, 5, and 7 on a 3-grade scale (I none or insignificant, II pain requiring no analgesic, II severe pain requiring analgesic) Trismus was measured on the day of operation and on days 2, 5, and 7 measuring the ability to open the mouth, using a vernier gauge Extraoral swelling was measured according to the method described by Lökken 1975. Dry socket diagnosis was made clinically on the basis of severe mandibular pain accompanied by necrotic debris or a denuded alveolus Wound healing (evidence of loose of periosteal flap and alveolitis) Side effects: patients were questioned at each examination regarding side effects such as fever, indisposition or diarrhoea
Notes	Only usable data that can be extracted by the paper are those on dry socket, subjects with no complications and adverse effects. Groups A and B have been considered together in the analysis All operations were carried out by the same surgeons under local anaesthesia

# Risk of bias

Bias	Authors' judgement	Support for judgement
Random sequence generation (selection bias)	Unclear risk	Quote: "Patients were assigned at random". Method of sequence generation not de- scribed
Allocation concealment (selection bias)	Unclear risk	Not reported.
Blinding of participants and personnel (performance bias) All outcomes	Low risk	Quote: "Double blind".
Blinding of outcome assessment (detection bias) All outcomes	Low risk	Quote: "Double blind".
Incomplete outcome data (attrition bias) All outcomes	Low risk	No missing data. All randomised partici- pants included in result analysis
Selective reporting (reporting bias)	Unclear risk	Data for swelling and trismus not reported, only comment that there was no difference
Other bias	Low risk	No other sources of bias identified.

Kaziro 1984

Methods	Study design: RCT. Conducted in: UK. Number of centres: 1. Recruitment period: not stated.
Participants	Inclusion criteria: patients with impacted mandibular wisdom teeth Exclusion criteria: not described. Age group: not stated. Number randomised: 118. Number evaluated: unclear.
Interventions	<b>Comparison: post-op metronidazole versus arnica versus placebo.</b> Group A (n = 41): metronidazole 400 mg 1 tablet twice daily post-operatively Group B (n = 39): arnica 200 tablets 1 tablet twice daily post-operatively Group C (n = 38): placebo 1 tablet twice daily post-operatively. Tablets were taken for 3 days. All participants had 2 Codis (aspirin plus codeine) tablets 3x daily for pain
Outcomes	Pain, trismus, oedema, wound healing on 4 <sup>th</sup> and 8 <sup>th</sup> post-op day, wound breakdown.
Notes	Data presented in graphs only. Extractions were done by 1 of 6 surgeons blinded to allocated treatment

# Risk of bias

Bias	Authors' judgement	Support for judgement
Random sequence generation (selection bias)	Unclear risk	Quotes: "randomised allocation" "ran- domly divided". Method of sequence gen- eration not described
Allocation concealment (selection bias)	Low risk	Code was kept by pharmacist at Royal Lon- don Homeopathic hospital
Blinding of participants and personnel (performance bias) All outcomes	Low risk	Quote: "Double blind".
Blinding of outcome assessment (detection bias) All outcomes	Low risk	Quote: "Double blind".
Incomplete outcome data (attrition bias) All outcomes	Unclear risk	Numbers of patients included in outcome assessments reported as percentage only
Selective reporting (reporting bias)	Unclear risk	All planned outcomes reported, but data only presented in graphs

Other bias	Unclear risk	No description of characteristics of patients by randomised group at baseline	
Mitchell 1986			
Methods	Study design: parallel group RCT. Conducted in: Newcastle, UK. Number of centres: 1. Recruitment period: not stated.		
Participants	Inclusion criteria: inpatients, aged 18-30 years, attending hospital for removal of 1 or more third molars Exclusion criteria: those with a significant medical history or acute infection were ex- cluded Age group: mean 24 years, range 17-33 years. Number randomised: 50 participants (89 teeth). Number evaluated: 50.		
Interventions	<b>Comparison: pre-op tinidazole versus pla</b> Group A (n = 25 participants, 45 teeth): tinid Group B (n = 25 participants, 44 teeth): pla All patients had ibuprofen as required while i after discharge	<b>acebo.</b> lazole 500 mg orally 12 hours pre-operatively acebo oral 12 hours pre-operatively n hospital and access to analgesics as required	
Outcomes	Infected socket, onset of painful socket incr signs of infection or dry socket, type of bon	easing in severity, within first 7 days. Other e removal	
Notes	4 surgeons conducted the extractions, using 1 clinician blinded to intervention assessed	a standardised technique all patients both pre- and post-operatively	
Risk of bias			

Bias	Authors' judgement	Support for judgement
Random sequence generation (selection bias)	Low risk	Quote: "allocated in accordance with a pre-determined randomisation code dur- ing pre-operative assessment"
Allocation concealment (selection bias)	Low risk	Quote: "drugs were individually packaged and allocated" -assumed allocation oc- curred at the pharmacy
Blinding of participants and personnel (performance bias) All outcomes	Low risk	Quote: "Double blind".

# Mitchell 1986 (Continued)

Blinding of outcome assessment (detection bias) All outcomes	Low risk	Quote: "Double blind".
Incomplete outcome data (attrition bias) All outcomes	Low risk	All randomised participants included in the outcome assessment
Selective reporting (reporting bias)	Unclear risk	Infection per socket and per patient re- ported. Pain swelling and trismus not re- ported
Other bias	Low risk	No other sources of bias identified.

# Barclay 1987

Methods	Study design: RCT. Conducted in: New Zealand. Number of centres: 1. Recruitment period: unclear. Funding source: metronidazole and placebo tablets were supplied by May and Baker New Zealand Ltd
Participants	Inclusion criteria: "patients with a history of non-acute pericoronitis, and therefore likely to experience a high prevalence of dry socket". Patients had to meet 2 or more of the following criteria: a history of 2 or more episodes of previously diagnosed pericoronitis; the expression of pus from beneath a pericoronal flap in the absence of significant symp- toms; radiographic enlargement of the follicular space distal to the third molar in the absence of significant symptoms; crater-like radiographic defect as described by Howe (Howe 1985). Exclusion criteria: pregnancy. Age group: mean 23 years, range 16-48 years. Group A: randomised 50; analysed 45. Group B: randomised 50; analysed 50.
Interventions	<b>Comparison: pre + post-op metronidazole versus placebo.</b> Group A: metronidazole 400 mg 1 hour before the intervention and then 3 times a day for 8 times Group B: placebo 1 hour before the intervention and then 3 times a day for 8 times All patients were given the same post-operative instructions, and were given 6 analgesic tablets (codeine phosphate and paracetamol)
Outcomes	Dry socket: continuous dull pain from an empty, or partially empty socket, or from the region of the socket. Pain: marked by patient on a 10 mm line (VAS). Outcomes were recorded at 2 and 7 days post-operatively
Notes	
Risk of bias	

# Barclay 1987 (Continued)

Bias	Authors' judgement	Support for judgement
Random sequence generation (selection bias)	Low risk	Quote: "Assigned to one of two groups by a table of random numbers"
Allocation concealment (selection bias)	Unclear risk	Not reported.
Blinding of participants and personnel (performance bias) All outcomes	Low risk	Double blind. Quote: "None of the pa- tients, nor the several operators, were aware of the active or placebo nature of the indi- vidual medication"
Blinding of outcome assessment (detection bias) All outcomes	Low risk	Double blind. Quote: "None of the pa- tients, nor the several operators, were aware of the active or placebo nature of the indi- vidual medication"
Incomplete outcome data (attrition bias) All outcomes	High risk	10% of antibiotic group not included in the analysis.
Selective reporting (reporting bias)	Low risk	Planned outcomes of dry socket, pain (VAS), adverse effects and compliance were reported
Other bias	Low risk	No other sources of bias identified.

# Happonen 1990

Methods	Study design: 3-arm RCT. Conducted in: Finland. Number of centres: 1. Recruitment period: unclear. Funding source: unclear.
Participants	Inclusion criteria: healthy consecutive students seeking treatment for impacted, not on any drugs, with the exception of oral contraceptives Exclusion criteria: hypersensitivity to penicillin or codeine Age group: mean 24 years. Group A: randomised unclear; analysed 44. Group B: randomised unclear; analysed 47. Group C: randomised unclear; analysed 45. 8 of the patients enrolled (total 144) were not included in the analysis, but it is unclear in which group they were allocated
Interventions	<b>Comparison: pre- + post-op penicillin versus pre- + post-op tinidazole versus placebo.</b> Group A: 1 tablet of phenoxmethylpenicillin 660 mg 1 hour before operation and then 3 times a day for 14 times

# Happonen 1990 (Continued)

	Group B: 1 tablet of tinidazole 500 mg 1 hour before operation and then 3 times a day for 14 times
	Group C: 1 tablet of placebo 1 hour before operation and then 3 times a day for 14 times A 1 minute mouth rinse of 0.2% chlorhexidine was given before surgery 3 tablets of a preparation containing aminophenazon (300 mg), phenobarbital (50 mg) codeine (30 mg) and caffeine (100 mg) was provided to all participants as a rescue analgesic to be taken when needed
Outcomes	Time of onset and resolution of post-operative swelling, as well as time of maximum swelling, as recorded by patients Post-operative pain every hour during the day of the surgery, and at intervals of 4 and 6 hours on the first and second post-operative day respectively. Number of analgesics was also reported Maximal opening of the mouth was measured before and after surgery (sixth day) Patients were visited on the sixth post-operative day and signs of infection, fever, swelling and tender lymph nodes were recorded by the clinicians
Notes	Group A and B were considered together in the present review All operations were carried out under local anaesthesia, by 1 surgeon, using a standardised procedure, 1 tooth being operated at a time

# Risk of bias

Bias	Authors' judgement	Support for judgement
Random sequence generation (selection bias)	Unclear risk	Quote: "randomly assigned". Method of se- quence generation not described
Allocation concealment (selection bias)	Unclear risk	Not mentioned.
Blinding of participants and personnel (performance bias) All outcomes	Low risk	Quote: "Double blind".
Blinding of outcome assessment (detection bias) All outcomes	Low risk	Quote: "Double blind".
Incomplete outcome data (attrition bias) All outcomes	High risk	8 out of 144 patients were lost at follow- up (5%), and it is unclear which groups these were from. No specific ITT approach is adopted
Selective reporting (reporting bias)	High risk	Planned outcomes of duration of swelling, infection, fever reported. Pain (VAS) re- ported only in graph for first 13 hours, no data at day 7, yet this was the main reason given for time off work
## Happonen 1990 (Continued)

Other bias	Low risk	No other sources of bias identified.	
Ritzau 1992			
Methods	Study design: RCT. Conducted in: Denmark. Number of centres: 2. Recruitment period: betwee Funding source: unclear.	Study design: RCT. Conducted in: Denmark. Number of centres: 2. Recruitment period: between October 1987 and November 1988. Funding source: unclear.	
Participants	Inclusion criteria: healthy s tially or totally) mandibular Exclusion criteria: any me pericoronitis, patients who also excluded Age group: not stated. A total of 312 subjects were Group A: randomised uncle Group B: randomised uncle 42 subjects did not complet voluntarily, 1 had intercurre did not present for surgery	Inclusion criteria: healthy subjects scheduled for surgical removal of an impacted (par- tially or totally) mandibular third molar Exclusion criteria: any medical condition that might interfere with the study, acute pericoronitis, patients who had taken antibiotics within 48 hours before surgery were also excluded Age group: not stated. A total of 312 subjects were randomised in 2 groups. Group A: randomised unclear; analysed 135. Group B: randomised unclear; analysed 135. 42 subjects did not complete the study: 4 did not comply with the protocol, 4 withdrew voluntarily, 1 had intercurrent disease, 11 were lost to follow-up for various reasons, 22 did not present for surgery after have been enrolled	
Interventions	<b>Comparison: pre-op metr</b> Group A: 2 tablets with a before surgery Group B: 2 tablets of place	<b>onidazole versus placebo.</b> total dose of 1000 mg metronidazole no later than 30 min bo no later than 30 min before surgery	
Outcomes	Follow- up examination wa be removed. Alveolitis sicca simultaneously present: 1 ipsilateral ear, and 2 disinte	as scheduled for a week after surgery when sutures were to a dolorosa (dry socket) was diagnosed when 2 criteria were severe pain irradiating from the empty socket towards the gration (partial or total) of the socket coagulum	
Notes			

Risk of bias

Bias	Authors' judgement	Support for judgement
Random sequence generation (selection bias)	Low risk	Quote: "random sequence generated by a computerized program"
Allocation concealment (selection bias)	Low risk	Quote: "the code was unknown to the in- vestigators until the termination of collec- tion of clinical data"

Blinding of participants and personnel (performance bias) All outcomes	Low risk	Quote: "double-blind" and "metronida- zole and placebo were manufactured in the shape of pills of identical size, shape, weight, and colour, packed and code num- bered"
Blinding of outcome assessment (detection bias) All outcomes	Low risk	Quote: "double-blind" and "metronida- zole and placebo were manufactured in the shape of pills of identical size, shape, weight, and colour, packed and code num- bered"
Incomplete outcome data (attrition bias) All outcomes	High risk	22/312 randomised participants did not have surgery. 20/290 (7%) patients who did undergo surgery were excluded from the outcome evaluation, but allocated treat- ment not stated. No specific ITT approach is reported, attrition rate is higher than event rate (4.8%) and bias in these results is considered to be likely
Selective reporting (reporting bias)	Unclear risk	Methods section states that planned out- come was alveolitis sicca dolorosa (ASD) and this was reported, but other clinically important outcomes were not reported
Other bias	Unclear risk	13% loss of participants post-randomi- sation and no baseline characteristics re- ported for each group. Biases possible

Leon Arcila 20	01
----------------	----

Methods	Study design: RCT. Conducted in: University of Valley, Colombia. Number of centres: 1. Recruitment period: 1 September 1998 to 1 September 2000.
Participants	Inclusion criteria: patients aged 14-53 years, ASA1, with good oral hygiene, bacterial plaque index = 30%, no oral cavity infections or inflammation or pericoronitis, who<br required extraction of third molars Exclusion criteria: allergy to penicillin. Age group: not stated. Number randomised: 102. Number evaluated:102.
Interventions	<b>Comparison: pre- + post-op amoxicillin versus placebo.</b> Group A (n = 49): amoxicillin 1 gm orally 1 hour pre-operatively and 6 hours post-operatively

#### Leon Arcila 2001 (Continued)

	Group B (n = 53): placebo 1 hour pre-op and 6 hours post-operatively
Outcomes	Infection.
Notes	All patients had a single extraction - 38 upper teeth and 64 lower teeth Additional information supplied by author in response to email request

## Risk of bias

Bias	Authors' judgement	Support for judgement
Random sequence generation (selection bias)	Low risk	Quote: "Participants were randomised us- ing a computer" (email from author)
Allocation concealment (selection bias)	Low risk	Quote: "One of the researchers allocated the treatment. Surgeon, patient and statis- tician did not know such information" (email from author)
Blinding of participants and personnel (performance bias) All outcomes	Low risk	Quote: "Double blind".
Blinding of outcome assessment (detection bias) All outcomes	Low risk	Quote: "Double blind".
Incomplete outcome data (attrition bias) All outcomes	Low risk	Quote: "No drop outs or losses to follow up. Everybody was included" (email from author)
Selective reporting (reporting bias)	Unclear risk	Infection was the only planned outcome.
Other bias	Low risk	No other sources of bias identified.

## Sekhar 2001

Methods	Study design: RCT. Conducted in: India. Number of centres: 1. Recruitment period: not stated. Funding source: not stated.
Participants	Inclusion criteria: patients aged 19-36 requiring removal of lower wisdom teeth under local anaesthesia Exclusion criteria: pre-existing abscess or cellulitis, acute pericoronitis, pre-existing con- ditions associated with third molars, xerostomia. Those requiring antibiotic prophylaxis for other reasons, immunocompromised patients, pregnancy, diabetes, cancer or renal

## Sekhar 2001 (Continued)

	failure and those who had received antibiotics in 2 weeks prior to start of study Age group: mean 30 years. Number randomised: 151 (53, 61 & 37 in Groups A, B, C respectively) Number evaluated: 125 (44, 47 & 34 in Groups A, B, C respectively)
Interventions	<b>Comparison: pre-op versus post-op metronidazole versus placebo.</b> Group A: metronidazole 1 g 1 hour pre-op. Group B: metronidazole 400 mg 8 hourly for 5 days. Group C: placebo - frequency of administration not specified All participants had a prescription for ibuprofen 400 mg to be taken as required for pain relief
Outcomes	Pain (4 point scale) measured on days 2 and 6 post-op, inter-incisal mouth opening (mm) whether there was purulent discharge from wound, dry socket on day 6
Notes	Surgeons performing the extractions were either consultants, post-graduate trainees or house officers

#### Risk of bias

Bias	Authors' judgement	Support for judgement
Random sequence generation (selection bias)	Unclear risk	Quote: "randomly assigned using pre- pared randomizations in sealed envelopes". Method of sequence generation not de- scribed
Allocation concealment (selection bias)	Low risk	Allocation concealed in sealed envelopes.
Blinding of participants and personnel (performance bias) All outcomes	Low risk	Described as double blind, but dosing schedule different in each group. Outcome assessor was blinded to allocated treatment
Blinding of outcome assessment (detection bias) All outcomes	Low risk	Described as double blind, but dosing schedule different in each group. Outcome assessor was blinded to allocated treatment
Incomplete outcome data (attrition bias) All outcomes	High risk	26/151 (17%) (9, 14 and 3 from groups A, B and C) of those randomised were ex- cluded because they did not return for fol- low-up evaluation. Those excluded were more likely to have had bone removed and had longer mean operating times. Given low event rate this attrition could have re- sulted in biased outcome estimates
Selective reporting (reporting bias)	Low risk	All planned outcomes reported.

## Sekhar 2001 (Continued)

Other bias	Unclear risk	Percentage of patients in 2x daily metron- idazole group who had bone removed ap- pears to be significantly lower compared to other groups
Bergdahl 2004		
Methods	Study design: RCT. Conducted in: Sweden. Number of centres: 1. Recruitment period: unclear. Funding source: unspecified.	
Participants	Inclusion criteria: healthy subjects, not taking any other drugs apart from oral contra- ceptives, who needed removal of unilateral or bilateral mandibular third molar teeth. Only partially impacted teeth, which had partly broken through the mucosa, with a communication to the oral cavity, requiring surgical flap, were included in the study Exclusion criteria: subjects with teeth completely covered with mucosa Age group: mean 29 years, range 17-65 years. Group A: randomised 60; analysed 59. Group B: randomised 60; analysed 60.	
Interventions	<b>Comparison: pre-op metronidazole versus placebo.</b> Group A: metronidazole 1600 mg as a single dose 45 min before the intervention Group B: placebo as a single dose 45 min before the intervention All patients were given the same post-operative instructions, and were given 20 analgesic tablets (paracetamol 500 mg with codeine 30 mg)	
Outcomes	Dry socket assessed 4 days post-op.	
Notes	Patients with acute pericoronitis were opera toms of pericoronitis had ceased Sample size calculation reported.	ated on after objective and subjective symp-

Risk of bias

Bias	Authors' judgement	Support for judgement
Random sequence generation (selection bias)	Unclear risk	Quote: "a randomised trial". Method of se- quence generation not described
Allocation concealment (selection bias)	Unclear risk	Not mentioned.
Blinding of participants and personnel (performance bias) All outcomes	Low risk	Quote: "double blind".

## Bergdahl 2004 (Continued)

Blinding of outcome assessment (detection bias) All outcomes	Low risk	Quote: "double blind".
Incomplete outcome data (attrition bias) All outcomes	Low risk	Only "one patient had to be withdrawn be- cause he had taken an oral antibiotic for other reasons two days after operation"
Selective reporting (reporting bias)	High risk	Pain, bad odour or taste as assessed by pa- tients were not reported
Other bias	Unclear risk	Short duration of follow-up (4 days), un- clear whether patients with acute pericoro- nitis prior to trial were treated with antibi- otics

#### Arteagoitia 2005

Methods	Study design: RCT. Conducted in: Spain. Number of centres: 1. Recruitment period: between March 2001 and February 2003. Funding source: financed by the Health Research Fund FIS/GRAN dossier number 00/ 0585. The trial patients' insurance was taken out by the Basque Health Department, Basque Health Service/Osakidetza, Osakidetza, pursuant to the conditions laid down in RD 561/1993. The antibiotic and placebo were supplied free of charge by Géminis (Novartis generics). Chlorhexidine was supplied free of charge by LACER
Participants	Inclusion criteria: patients needing a third molar extraction under local anaesthesia Exclusion criteria: patients with any bacterial endocarditis risk factors, pregnant and breastfeeding women, patients with acute infections 10 days prior to the intervention, those who had to take antibiotics and those with a history of allergy or intolerance to the drugs used Age group: mean 24 years, range 18-60 years. Group A: randomised 233; analysed 233 (ITT analysis). Group B: randomised 261; analysed 261 (ITT analysis).
Interventions	<b>Comparison: post-op amoxicillin/clavulanate versus placebo.</b> Group A: amoxicillin/clavulanic acid 500/125 mg oral 3 times a day for 4 days after the procedure Group B: placebo oral 3 times a day for 4 days after the procedure All patients had irrigation of the alveolus with 0.12% chlorhexidine digluconate, and chlorhexidine mouthwashes were used for 3 days
Outcomes	Fever (oral temperature >37.88 after 24 hours for no other justifiable cause); intraoral abscess diagnosed via fluctuation pus drainage; dry socket defined as absence of clot with necrotic remains present in the alveolus accompanied by severe mandibular pain; severe pain persisting or increasing 48 hours after surgery accompanied by intraoral in-

## Arteagoitia 2005 (Continued)

	flammation (moderate or severe) and/or intraoral erythema (moderate or severe); severe pain after 7th day accompanied by intraoral inflammation (moderate or severe) and/or intraoral erythema (moderate or severe) for no other justifiable reason which improves with antibiotic treatment. Lack of inflammatory complications. Diagnosis of post-oper- ative infection and inflammatory complication was performed by the main researcher, according to previously published clinical criteria
Notes	All extractions were performed by maxillofacial surgeons, under locoregional anaesthetic of the inferior alveolar and buccal nerves with Ultracain

## Risk of bias

Bias	Authors' judgement	Support for judgement
Random sequence generation (selection bias)	Low risk	Quote: "The simple randomizations was performed using C4-SDP program and , which was used as the patient's number"
Allocation concealment (selection bias)	Low risk	Quote: "Each of enrolled patients was as- signed the corresponding blinded random successive treatment number"
Blinding of participants and personnel (performance bias) All outcomes	Low risk	Quote: "Double-blinded".
Blinding of outcome assessment (detection bias) All outcomes	Low risk	Quote: "Double-blinded".
Incomplete outcome data (attrition bias) All outcomes	Low risk	2 participants lost to follow-up from each group but intention-to-treat analysis was performed
Selective reporting (reporting bias)	High risk	Planned outcomes of pain inflammation and erythema measured qualitatively and reported but VAS pain scores measured and not reported
Other bias	Low risk	No other sources of bias identified.

Halpern 2007

Methods	Study design: RCT. Conducted in: US. Number of centres: 1. Recruitment period: between 1 June 2002 and 1 July 2005. Funding source: supported in part by the Oral and Maxillofacial Surgery Foundation Research Grant and Massachusetts General Hospital (MGH) Center for Applied Clinical Investigation	
Participants	Inclusion criteria: patients needing a third molar extraction under intravenous sedation or general anaesthesia in the office-based ambulatory setting Exclusion criteria: subjects with pre-existing conditions that could affect wound healing or predispose them to inflammatory complications, including previous radiation therapy to the maxillofacial region, human immunodeficiency virus infection, organ or marrow transplant candidates or recipients, diabetes, or organ failure (kidney, heart, liver); sub- jects requiring antibiotic prophylaxis for endocarditis, were currently on oral steroid ther- apy, were allergic to the antibiotics proposed for use in this study, deferred intravenous sedation or general anaesthesia; had local pathology, e.g. cysts or tumour, associated with M3s that was not incidental to the removal of the M3; acute inflammation in the area of the planned extraction characterized by frank purulence, erythema, induration, or trismus; supernumerary teeth to be removed; or deferred study participation Age group: mean 25 years. Group A: randomised 60; analysed 59. Group B: randomised 62; analysed 59.	
Interventions	<b>Comparison: pre-op IV penicillin (or clindamycin) versus placebo.</b> Group A: solution of penicillin (15,000 units per kilogram) or, for penicillin-allergic subjects, clindamycin (600 mg) administered intravenously within 1 hour before the intervention Group B: placebo solution (10 cc saline 0.9%) administered intravenously within 1 hour before the intervention Post-operative analgesia consisted of the use of 1 or 2 acetaminophen (500 mg) and hydrocodone (5 mg) tablets administered orally every 3 to 4 hours	
Outcomes	Dry socket (a new onset or increasing pain more than 36 hours after the operation, with a loss of the blood clot in the extraction site as evidenced by exposed bone, gentle probing or irrigation of the wound duplicating the pain, and significant pain relief after application of an anodyne dressing; all elements needed to be present to make the diagnosis) Surgical site infection (visual evidence of frank purulence in one or more of the extraction sites and a Gram's stain demonstrating white blood cells present) Any post-operative inflammatory complications (dry socket or surgical site infection) Assessed on day 7 post-operatively (range 5-14).	
Notes		
Risk of bias		
Bias	Authors' judgement	Support for judgement

## Halpern 2007 (Continued)

Random sequence generation (selection bias)	Unclear risk	Quote: "randomized". Method of sequence generation not described
Allocation concealment (selection bias)	Low risk	Quote: "Consecutively numbered, double- sealed envelopes were prepared containing the treatment assignment"
Blinding of participants and personnel (performance bias) All outcomes	Low risk	Quote: "Double blind. The surgeon and study subject were blinded to the true na- ture of the contents of the syringe"
Blinding of outcome assessment (detection bias) All outcomes	Low risk	Quote: "Double blind. The surgeon and study subject were blinded to the true na- ture of the contents of the syringe"
Incomplete outcome data (attrition bias) All outcomes	Unclear risk	1/60 and 3/62 participants lost to follow- up in the antibiotic and placebo groups, but low event rates mean that data from these participants could have changed the outcome
Selective reporting (reporting bias)	Unclear risk	Surgical site infections and acute osteitis planned and reported. No report of pain swelling or trismus
Other bias	Low risk	No other sources of bias identified.

## Kaczmarzyk 2007

Methods	Study design: 3-arm RCT. Conducted in: Poland. Number of centres: 1. Recruitment period: between January 2005 and April 2006. Funding source: unclear.
Participants	Inclusion criteria: healthy volunteers, needing surgical extraction of a retained lower third molar, which was not the cause of inflammation (mainly due to orthodontic rec- ommendations) that required bone removal Exclusion age under 18 or over 60, pregnancy, allergy to clindamycin, lactose intolerance (lactose was the main component of the placebo), episodes of diarrhoea after antibiotic therapy in the interview, any digestive diseases, inflammation in the area of the tooth to be extracted, and any antibiotic or analgesic intake within the previous 7 days Age group: mean 24 years. Group A: randomised unclear; analysed 31. Group B: randomised unclear; analysed 28. Group C: randomised unclear; analysed 27. Of the 100 patients enrolled 9 did not check in for the follow-up examination, 3 were disqualified due to complications and 2 resigned during the trial without stating any

## Kaczmarzyk 2007 (Continued)

	reason
Interventions	<b>Comparison: pre-op versus pre- + post-op clindamycin versus placebo.</b> Group A: single-dose group: patients receiving 600 mg clindamycin hydrochloride orally 60 min pre-operatively, followed by a 300 mg placebo every 8 hours for 5 days Group B: 5-day group: patients receiving 600 mg clindamycin hydrochloride orally 60 min pre-operatively, followed by a dose of 300 mg clindamycin hydrochloride every 8 hours for 5 days Group C: placebo group: patients receiving 600 mg placebo orally 60 min prior to surgery, followed by a dose of 300 mg placebo every 8 hours for 5 days Only groups B and C were considered for the present review.
Outcomes	On the first, second and seventh post-operative day the following outcomes were eval- uated: trismus (on a 4-grade scale), facial swelling (on a 4-grade scale), submandibular lymphadenopathy (on a 4-grade scale), body temperature, pain (on a 100-mm VAS) , alveolar osteitis (clinical diagnosis of this complication was given in the case of the presence of a necrotic grey clot in a bare bony socket, foetor ex ore, accompanied by pain in this area)

Notes

## Risk of bias

Bias	Authors' judgement	Support for judgement
Random sequence generation (selection bias)	Low risk	Quote: "group assignment for one patient, determined in advance by a random num- ber table"
Allocation concealment (selection bias)	Low risk	Quote: "One hundred opaque and sequen- tially numbered envelopes were used for the concealment of allocation to trial groups"
Blinding of participants and personnel (performance bias) All outcomes	Low risk	Quote: "The patients, the surgeon per- forming the qualification, operative proce- dure and follow-up examination, and the statistician were not aware of who received which study intervention"
Blinding of outcome assessment (detection bias) All outcomes	Low risk	Quote: "The patients, the surgeon per- forming the qualification, operative proce- dure and follow-up examination, and the statistician were not aware of who received which study intervention"
Incomplete outcome data (attrition bias) All outcomes	High risk	14 out of 100 patients were lost at fol- low-up (14%). No specific ITT approach is adopted and it is unclear which groups these were from. Due to low event rates

# Kaczmarzyk 2007 (Continued)

		for the dichotomous outcomes incomplete outcome data could have resulted in bias	
Selective reporting (reporting bias)	Low risk	Planned outcomes of post-operative in- flammation (swelling, lymphadenopathy, trismus), pain, body temperature, and alve- olar osteitis reported	
Other bias	Low risk	No other sources of bias identified.	
Lacasa 2007			
Methods	Study design: 3-arm RCT. Conducted in: Spain. Number of centres: 1. Recruitment period: between January and Funding source: the trial was supported I Cantos, Madrid, Spain	Study design: 3-arm RCT. Conducted in: Spain. Number of centres: 1. Recruitment period: between January and December 2002. Funding source: the trial was supported by a grant from GlaxoSmithKline S.A., Tres Cantos, Madrid, Spain	
Participants	Inclusion criteria: adult patients (>18 years of age) who were going to have third mandibu- lar molar surgery Exclusion criteria: patients were excluded if they had a recent local infection prior to surgery (<15 days), had known or suspected allergy to beta-lactams, known or suspected allergy to metamizol, history of renal failure, blood dyscrasia or chronic liver disease of any type, antecedents of recent and/or symptomatic peptic ulcer, or were on antiaggregants or corticosteroids prior (<15 days) to entry. Female patients of child-bearing potential had to have a negative urine pregnancy test prior to enrolment Age group: mean 29 years. Group A: randomised 75; analysed (day 7) 62. Group B: randomised 75; analysed (day 7) 68. Group C: randomised 75; analysed (day 7) 69.		
Interventions	<b>Comparison: pre-op versus post-op amoxicillin/clavulanate versus placebo.</b> Group A: 2 amoxicillin/clavulanate 1000/62.5 mg matching placebo tablets (2000/ 125 mg) in a single dose before surgery, plus 2 amoxicillin/clavulanate 1000/62.5 mg matching placebo tablets (2000/125 mg, BID) for 5 days Group B: 2 active amoxicillin/clavulanate 1000/62.5 mg tablets (2000/125 mg) in a single dose before surgery followed by 2 matching placebo tablets (2000/125 mg, BID) for 5 days Group C: 2 matching placebo tablets (2000/125 mg) in a single dose before surgery followed by 2 active amoxicillin/clavulanate 1000/62.5 mg tablets (2000/125 mg, BID) for 5 days All patients were matched to receive the same analgesic drug throughout the study period with identical dosage. Metamizol (NolotilTM capsules) was used, 1 capsule every 8 hours, for a minimum of 48 hours, since it is much less anti-inflammatory than other analgesics. Patients were able to continue receiving analgesia afterwards (according to the investigator's judgement), depending on the presence of pain		

## Lacasa 2007 (Continued)

Outcomes	The main study variables and subjective well being were evaluated on days 1, 3 and 7 Infection was defined by any of the following: (1) presence of purulent discharge in the extraction socket and/or excessive swelling with fluctuation, with or without pain; (2) presence of a local abscess; (3) onset of facial or cervical cellulitis plus other signs suggesting infection such as pain, increased heat, erythema and/or fever; (4) presence of osteitis of dental alveolus defined as absence of the haematic clot of the orifice and presence of a putrid smell and intense neuralgic type pain Other inflammatory outcomes were recorded individually, and in a composite way using an inflammation score tabular display with a maximum permitted score of 10. They included swelling, trismus, pain, dysphagia, fever
Notes	Only groups A and C were considered for the present review. 2 of the authors are employees of the funding company.

Risk of bias

Bias	Authors' judgement	Support for judgement
Random sequence generation (selection bias)	Unclear risk	Quote: "randomised". Method of sequence generation not described
Allocation concealment (selection bias)	Unclear risk	Not mentioned.
Blinding of participants and personnel (performance bias) All outcomes	Low risk	Quote: "Double blind".
Blinding of outcome assessment (detection bias) All outcomes	Low risk	Quote: "Double blind".
Incomplete outcome data (attrition bias) All outcomes	High risk	It is unclear whether authors used any ITT analysis. 3/225, 9/225 and 26/225 partici- pants are lost to follow-up at days 1, 3 and 7 respectively. Given the low rate of infec- tion this attrition could have introduced a bias
Selective reporting (reporting bias)	High risk	According to the methods the planned outcomes were infection, inflammation, swelling, trismus, pain, dysphagia, fever and adverse effects. Data are reported for infection, and means without variance esti- mates for pain, but no other outcome data reported
Other bias	Unclear risk	Statistically significant difference in dura- tion of operation between the placebo and

		pre-emptive groups. 2 of the authors are employed by the company that funded the trial
Bezerra 2011		
Methods	Study design: RCT cross-over. Conducted in: Brazil. Number of centres: 1. Recruitment period: January to Novembe	er 2008.
Participants	Inclusion criteria: healthy patients with a third molars, with similar degrees of impa Exclusion criteria: tobacco use, orthodonti feeding, chronic systemic disorders, allergi antibiotics and use of antibiotics in 3 mo Age group: mean 21 years, range 18-31 y Number randomised: 36. Number evaluated: 34.	no periodontal disease requiring removal of 4 action between sides of mouth ic bands on second molars, pregnancy or breast- ies to antibiotics, history of adverse effects from nths prior to entering trial ears.
Interventions	<b>Comparison: pre-op amoxicillin versus</b> Group A: amoxicillin 2 x 500 mg admini Group B: placebo (2 tablets) identical in Standard post-operative treatment was N 4 days and dipyrone (NSAID) 500 mg 6	<b>placebo.</b> stered orally 1 hour pre-op appearance administered 1 hour pre-op imesulid (NSAID) 100 mg every 12 hours for hourly for 2 days
Outcomes	Soft tissue edema/ulcer, pain (1-10 VAS), (purulent secretion, alveolitis (pain + par and 14 days post-op	edema, limitation of mouth opening, infection rtially/totally disintegrated clot), fever at 3, 7
Notes	Email from author 13/2/2012 stating du ditional outcome data provided	ration of washout period at least 45 days. Ad-
Pick of high		

#### Risk of bias

Bias	Authors' judgement	Support for judgement
Random sequence generation (selection bias)	Low risk	Drugs/placebo placed into transparent, sterile boxes with code number. Patient chose one box for first procedure and a coin toss decided which side of mouth was done first
Allocation concealment (selection bias)	Low risk	Unclear who performed the coin toss and how the result was communicated to the surgeon. However bias unlikely to result from this design

Antibiotics to prevent complications following tooth extractions (Review) Copyright © 2013 The Cochrane Collaboration. Published by John Wiley & Sons, Ltd. - -

.

## Bezerra 2011 (Continued)

Blinding of participants and personnel (performance bias) All outcomes	Low risk	Double blind - neither patient nor surgeon knew which treatment was given
Blinding of outcome assessment (detection bias) All outcomes	Low risk	Double blind - neither patient nor surgeon knew which treatment was given
Incomplete outcome data (attrition bias) All outcomes	Low risk	2/36 patients were not included in analysis. Due to low number and cross-over design, attrition bias is unlikely
Selective reporting (reporting bias)	Low risk	Duration of surgery, inflammatory/infec- tious events, pain scores, edema, maximum mouth opening, swelling of soft tissues, alveolitis reported
Other bias	High risk	Indications of a significant period effect with regard to the outcome of pain. This is likely to bias this outcome towards no dif- ference between active and placebo
López-Cedrún 2011		
Methods	Study design: RCT. Conducted in: Spain. Number of centres: 1. Recruitment period: not stated.	
Participants	Inclusion criteria: at least 1 mandibular impacted or partially erupted third molar re- quiring extraction Exclusion criteria: aged >60 or <18 years, infectious or systemic diseases, immunosup- pressive treatment, smoking, peptic ulcer, pregnancy, lactation, known or suspected al- lergy to ibuprofen or beta-lactam antibiotics, carious or non-impacted third molars, pericoronitis, or patients in whom excessive technical difficulty was expected Age group: mean 22 years, range 18-46 years. Number randomised: 134. Number evaluated:123.	
Interventions	<b>Comparison: pre-op versus post-op amo</b> Group A: amoxicillin 4x 500 mg 2 hours p 3x daily for 5 days Group B: 4 placebo tablets 2 hours pre-op days	<b>xicillin versus placebo.</b> rior to surgery plus 15 placebo tablets taken plus 15 placebo tablets taken 3x daily for 5

# López-Cedrún 2011 (Continued)

Outcomes	Intraoral swelling, maximal mouth opening, pain (100 point VAS) dysphagia, fever, purulent wound discharge, alveolar osteitis (dry socket), side effects of treatment at 7 days post-op
Notes	All procedures were performed by the same surgeon. Additional information supplied by author in response to email request

# Risk of bias

Bias	Authors' judgement	Support for judgement
Random sequence generation (selection bias)	Low risk	Pharmacist held the randomisation code and the drug code (a random alpha numeric code)
Allocation concealment (selection bias)	Low risk	Quote: "A set of opaque sealed envelopes contained the drug code for every patient. One envelope was opened for every pa- tient and the patient was provided with the coded tablet pack that matched the num- ber in the envelope"
Blinding of participants and personnel (performance bias) All outcomes	Low risk	Quote: "Double blind".
Blinding of outcome assessment (detection bias) All outcomes	Low risk	Quote: "Double blind".
Incomplete outcome data (attrition bias) All outcomes	High risk	11/134 (8%) were excluded from the analy- sis. 3, 0, 1 were excluded from pre-op, post- op and placebo groups due to technical dif- ficulty of the procedure and 2,1, 4 due to inadequate follow-up. Due to the low event rate for infection it is probable that this at- trition introduced a bias to the outcome
Selective reporting (reporting bias)	Low risk	All planned outcomes reported.
Other bias	Unclear risk	Statistically significant difference in mean operating time between pre- and post-op antibiotic groups

Pasupathy 2011

Methods	Study design: RCT. Conducted in: India. Number of centres: 1. Recruitment period: unclear. Funding source: not stated.
Participants	Inclusion criteria: patients with mandibular mesioangularly impacted third molars re- quiring extraction Exclusion criteria: infections (space infections, acute pericoronitis), medically compro- mised, pregnant, allergic to either penicillin or metronidazole, those who have taken antibiotics in the 2 months prior to surgery Age group: mean 29 years, range 18-48 years. Number randomised: 98. Number evaluated: 89.
Interventions	<b>Comparison: pre-op amoxicillin versus pre-op metronidazole versus placebo.</b> Group A: amoxicillin 1 g orally 1 hour prior to surgery. Group B: metronidazole 800 mg orally 1 hour prior to surgery Group C: placebo. All patients received ibuprofen 600 mg 3x daily for pain.
Outcomes	Surgical wound infection, purulent discharge, fever, restricted mouth opening on day 7 post-op
Notes	Sample size: reported that estimated sample size required was 107 in each group. Trial recruited ~30 per group

Risk of bias

Bias	Authors' judgement	Support for judgement
Random sequence generation (selection bias)	Low risk	Quote: "randomization table was prepared using a software program and a random al- locating number was given to each patient"
Allocation concealment (selection bias)	Low risk	Quote: "Sealed envelopes with the allotted number were used and were dispensed by 1 of our post graduate trainees throughout the study according to the allotted random- ization number"
Blinding of participants and personnel (performance bias) All outcomes	Low risk	Triple blind - neither the patient nor the surgeon nor the outcome evaluator were aware of the allocated treatment
Blinding of outcome assessment (detection bias) All outcomes	Low risk	Triple blind - neither the patient nor the surgeon nor the outcome evaluator were aware of the allocated treatment

#### Pasupathy 2011 (Continued)

Incomplete outcome data (attrition bias) All outcomes	High risk	9/98 (9%) of randomised patients excluded from the analysis, due to either not return- ing for follow=up (n = 8) or use of antibi- otic (n = 1). Allocated treatment group not described for these 9. Given low event rate this attrition is likely to have introduced bias
Selective reporting (reporting bias)	High risk	Surgical wound infection, purulent dis- charge, reported for each group. Fever, pain and trismus is not reported per treatment group, although it seems likely that these data were collected.
Other bias	High risk	Trial is very clearly underpowered and this is likely to bias results towards the null hy- pothesis of no difference between interven- tions

ITT = intention-to-treat; RCT = randomised controlled trial; VAS = visual analogue scale.

# Characteristics of excluded studies [ordered by study ID]

Study	Reason for exclusion
Abu-Mowais 1990	Not double blind.
Ataoglu 2008	Not double blind.
Bargnesi 1985	Study of antibiotics used in conjunction with a range of small dental surgical procedures including but not limited to tooth extractions
Curran 1974	Described as double blind but control group received no treatment. Patients not blinded to treatment, and asked not to inform outcome assessors
de Moura 2011	Washout period 4 weeks (translated from Spanish).
Delilbasi 2004	Not double blind.
Foy 2004	Not double blind.
Fridrich 1990	Not randomised or quasi-randomised.
Graziani 2005	Not double blind.

## (Continued)

Grossi 2007	Not double blind.
Head 1984	Bacteraemia outcomes only.
Krekmanov 1980	Not double blind.
Krekmanov 1981	Not double blind.
Krekmanov 1986	Not double blind.
Laird 1972	This study compares 2 antibiotic regimens.
Limeres 2009	This study compares 2 antibiotic regimens.
Lombardia Garcia 1987	Not double blind.
Lopes 2011	Not double blind.
Luaces-Rey 2010	This study compares 2 antibiotic regimens.
Lyall 1991	Not double blind.
MacGregor 1973	Topical antibiotic.
Mitchell 1987	No blinding described.
Monaco 1999	Not double blind.
Monaco 2009	Not double blind.
Olusanya 2011	This study compares 2 antibiotic regimens.
Osborn 1979	From translator: "it is clear that this study is double blinded but it is unclear how participants were allocated to treatment groups. Random not mentioned"
Poeschl 2004	Not double blind.
Reekie 2006	Topical antibiotic therapy.
Rood 1979	Not randomised.
Samsudin 1994	Not randomised or quasi and not double blind.
Siddiqi 2010	Washout period only 3 weeks (communication from author).
Stavropoulos 2006	Authors considered only topical antibiotic therapy.
Sulejmanagie 2005	Not randomised or quasi-randomised and not double blind.

#### (Continued)

Swanson 1989	Topical antibiotic therapy.
Uluibau 2005	Not double blind.
Walkow 1995	Abstract only, no mention of blinding and no subsequent trial report found
Yoshii 2002	No blinding described.

# Characteristics of ongoing studies [ordered by study ID]

# Arteagoitia 2011

Trial name or title	Efficacy of amoxicillin/clavulanic acid 2000/125 mg in preventing infection after extraction of impacted mandibular third molar totally covered by bone (EUDRACT 2008-005663-34)
Methods	Double-blind placebo-controlled trial.
Participants	Patients with bony impactions of mandibular third molars undergoing extraction under locoregional anaes- thesia
Interventions	Amoxicillin/clavulanic acid 2000/125 mg 1 hour pre-op plus post-op twice daily for 4 days versus placebo. All patients received 0.2% chlorhexidine rinses
Outcomes	Infection.
Starting date	26 June 2009.
Contact information	Atreagoitia M Stomatology Department, University of the Basque Country, Leioa, Spain
Notes	

## DATA AND ANALYSES

## Comparison 1. Antibiotic versus placebo

Outcome or subgroup title	No. of studies	No. of participants	Statistical method	Effect size
1 Local sign of infection	10		Risk Ratio (Random, 95% CI)	0.29 [0.16, 0.50]
1.1 Pre-operative prophylaxis	7		Risk Ratio (Random, 95% CI)	0.29 [0.15, 0.54]
1.2 Post-operative prophylaxis	4		Risk Ratio (Random, 95% CI)	0.15 [0.07, 0.31]
1.3 Pre- and post-operative prophylaxis	2		Risk Ratio (Random, 95% CI)	1.09 [0.40, 2.94]
2 Dry socket	9		Risk Ratio (Random, 95% CI)	0.62 [0.41, 0.95]
2.1 Pre-operative prophylaxis	6		Risk Ratio (Random, 95% CI)	0.75 [0.42, 1.33]
2.2 Post-operative prophylaxis	2		Risk Ratio (Random, 95% CI)	0.18 [0.01, 3.70]
2.3 Pre- and post-operative prophylaxis	3		Risk Ratio (Random, 95% CI)	0.52 [0.27, 0.99]
3 Pain (dichotomous on 6th-7th day)	3	675	Risk Ratio (M-H, Random, 95% CI)	0.60 [0.32, 1.11]
3.1 Pre-operative prophylaxis	1	61	Risk Ratio (M-H, Random, 95% CI)	0.97 [0.53, 1.76]
3.2 Post-operative prophylaxis	2	554	Risk Ratio (M-H, Random, 95% CI)	0.51 [0.14, 1.82]
3.3 Pre- and post-operative prophylaxis	1	60	Risk Ratio (M-H, Random, 95% CI)	0.36 [0.13, 0.98]
4 Pain score (VAS 7th day)	4		Mean Difference (Random, 95% CI)	-8.17 [-11.90, -4.45]
4.1 Pre-operative prophylaxis	3		Mean Difference (Random, 95% CI)	-7.41 [-16.18, 1.36]
4.2 Pre- and post-operative prophylaxis	3		Mean Difference (Random, 95% CI)	-8.30 [-13.18, -3.42]
5 Fever (6th-7th day)	4	816	Risk Ratio (M-H, Random, 95% CI)	0.34 [0.06, 1.99]
6 Swelling (7th day)	3	334	Risk Ratio (M-H, Random, 95% CI)	0.92 [0.65, 1.30]
6.1 Pre-operative prophylaxis	3	165	Risk Ratio (M-H, Random, 95% CI)	1.13 [0.69, 1.83]
6.2 Post-operative prophylaxis	1	64	Risk Ratio (M-H, Random, 95% CI)	0.50 [0.24, 1.02]
6.3 Pre- and post-operative prophylaxis	2	105	Risk Ratio (M-H, Random, 95% CI)	1.07 [0.53, 2.17]
7 Trismus (7th day)	2	175	Risk Ratio (M-H, Random, 95% CI)	0.84 [0.42, 1.71]
8 Adverse effects	5	930	Risk Ratio (M-H, Random, 95% CI)	1.98 [1.10, 3.59]

## Analysis I.I. Comparison I Antibiotic versus placebo, Outcome I Local sign of infection.

Review: Antibiotics to prevent complications following tooth extractions

Comparison: I Antibiotic versus placebo

Outcome: I Local sign of infection

Study or subgroup	log [Risk Ratio] (SE)	Risk Ratio IV,Random,95% Cl	Weight	Risk Ratio IV,Random,95% CI
I Pre-operative prophylaxis				
Mitchell 1986	-1.2528 (0.4914)		18.2 %	0.29 [ 0.11, 0.75 ]
Sekhar 2001	0.1603 (1.6092)		2.9 %	1.17 [ 0.05, 27.50 ]
Halpern 2007 (1)	-2.3979 (1.4658)		3.5 %	0.09 [ 0.01, 1.61 ]
Lacasa 2007	-1.0854 (0.614)		14.0 %	0.34 [ 0.10, 1.13 ]
Bezerra 2011	0 (0)			Not estimable
L pez-Cedr n 2011	-2.5903 (1.4876)	<del>د ا</del>	3.4 %	0.07 [ 0.00, 1.38 ]
Pasupathy 2011	-1.1325 (0.8844)		8.3 %	0.32 [ 0.06, 1.82 ]
Subtotal (95% CI)		•	50.3 %	0.29 [ 0.15, 0.54 ]
Heterogeneity: Tau <sup>2</sup> = 0.0; Chi <sup>2</sup> Test for overall effect: Z = 3.83 2 Post-operative prophylaxis	= 2.28, df = 5 (P = 0.81); $I^2 = 0.0\%$ (P = 0.00013)			
Sekhar 2001	0 (0)			Not estimable
Arteagoitia 2005	-1.9062 (0.4745)		18.9 %	0.15 [ 0.06, 0.38 ]
Lacasa 2007	-1.7644 (0.791)		9.8 %	0.17 [ 0.04, 0.81 ]
L pez-Cedr n 2011	-2.7081 (1.4886)	·	3.4 %	0.07 [ 0.00, 1.23 ]
Subtotal (95% CI) Heterogeneity: Tau <sup>2</sup> = 0.0; Chi <sup>2</sup> Test for overall effect: Z = 4.91 3 Pre- and post-operative propion Happonen 1990	= 0.32, df = 2 (P = 0.85); l <sup>2</sup> =0.0% (P < 0.00001) hylaxis 0.0843 (0.5076)	◆ ◆	<b>32.1 %</b>	0.15 [ 0.07, 0.31 ]
Leon Arcila 2001	0 (0)			Not estimable
<b>Subtotal (95% CI)</b> Heterogeneity: not applicable Test for overall effect: Z = 0.17	(P = 0.87)	-	17.6 %	1.09 [ 0.40, 2.94 ]
<b>Total (95% CI)</b> Heterogeneity: Tau <sup>2</sup> = 0.21; Ch Test for overall effect: $Z = 4.37$ Test for subgroup differences: C	$i^2 = 12.46$ , df = 9 (P = 0.19); $i^2 = 28$ (P = 0.000013) (hi <sup>2</sup> = 9.86, df = 2 (P = 0.01), $i^2 = 86$	◆ % 	100.0 %	0.29 [ 0.16, 0.50 ]
	1	0.005 0.1 10 200 Favours antibiotic Favours placebo		

(1) Adminstered intravenously

## Analysis I.2. Comparison I Antibiotic versus placebo, Outcome 2 Dry socket.

Review: Antibiotics to prevent complications following tooth extractions

Comparison: I Antibiotic versus placebo

Outcome: 2 Dry socket

Study or subgroup	log [Risk Ratio]	Risk Ratio	Weight	Risk Ratio
	(SE)	IV,Random,95% Cl		IV,Random,95% CI
l Pre-operative prophylaxis				
Ritzau 1992	-0.1542 (0.5429)		15.9 %	0.86 [ 0.30, 2.48 ]
Bergdahl 2004	-0.2456 (0.3786)		32.8 %	0.78 [ 0.37, 1.64 ]
Kaczmarzyk 2007	-1.5622 (1.1793)		3.4 %	0.21 [ 0.02, 2.12 ]
Halpern 2007 (I)	0 (0)			Not estimable
L pez-Cedr n 2011	0 (0)			Not estimable
Bezerra 2011	0 (1.4044)		2.4 %	1.00 [ 0.06, 15.68 ]
Subtotal (95% CI)		•	54.5 %	0.75 [ 0.42, 1.33 ]
Heterogeneity: Tau <sup>2</sup> = 0.0; Chi <sup>2</sup>	$P^2 = 1.28$ , df = 3 (P = 0.73); $I^2 = 0.05$	%		
Test for overall effect: $Z = 0.99$	(P = 0.32)			
2 Post-operative prophylaxis				
Arteagoitia 2005	-1.7234 (1.5466)		2.0 %	0.18 [ 0.01, 3.70 ]
L pez-Cedr n 2011	0 (0)			Not estimable
Subtotal (95% CI)			2.0 %	0.18 [ 0.01, 3.70 ]
Heterogeneity: not applicable				
Test for overall effect: $Z = 1.11$	(P = 0.27)			
3 Pre- and post-operative prop	hylaxis			
Bystedt 1981	-0.6931 (0.9618)		5.1 %	0.50 [ 0.08, 3.29 ]
Barclay 1987	-0.6484 (0.3763)		33.2 %	0.52 [ 0.25, 1.09 ]
Kaczmarzyk 2007	-0.6931 (0.9449)		5.3 %	0.50 [ 0.08, 3.19 ]
Subtotal (95% CI)		•	43.5 %	0.52 [ 0.27, 0.99 ]
Heterogeneity: Tau <sup>2</sup> = 0.0; Chi <sup>2</sup>	<sup>2</sup> = 0.00, df = 2 (P = 1.00); l <sup>2</sup> = 0.05	%		
Test for overall effect: $Z = 2.01$	(P = 0.045)			
Total (95% CI)		•	100.0 %	0.62 [ 0.41, 0.95 ]
Heterogeneity: $Tau^2 = 0.0$ ; Chi <sup>2</sup>	$P^2 = 2.65$ , df = 7 (P = 0.92); $I^2 = 0.09$	%		
Test for overall effect: $Z = 2.21$	(P = 0.027)			
Test for subgroup differences: C	$Chi^2 = 1.36$ , df = 2 (P = 0.51), $I^2 = 0$	).0%		
		0.01 0.1 1 10 100		

. . . . . . . .

Favours experimental Favours placebo

(1) Administered intravenoulsy

## Analysis I.3. Comparison I Antibiotic versus placebo, Outcome 3 Pain (dichotomous on 6th-7th day).

Review: Antibiotics to prevent complications following tooth extractions

Comparison: I Antibiotic versus placebo

Outcome: 3 Pain (dichotomous on 6th-7th day)

Study or subgroup	Antibiotic	Placebo	Risk Ratio	Weight	Risk Ratio
	n/N	n/N	H,Random,95% Cl		H,Random,95% Cl
l Pre-operative prophylaxis					
Sekhar 2001	20/44	8/17	+	31.7 %	0.97 [ 0.53, 1.76 ]
Subtotal (95% CI)	44	17	+	31.7 %	0.97 [ 0.53, 1.76 ]
Total events: 20 (Antibiotic),	8 (Placebo)				
Heterogeneity: not applicable	2				
Test for overall effect: $Z = 0$ .	(P = 0.9 )				
2 Post-operative prophylaxis					
Sekhar 2001	17/47	7/17		29.1 %	0.88 [ 0.44, 1.74 ]
Arteagoitia 2005	4/259	4/23		18.8 %	0.25 [ 0.09, 0.76 ]
Subtotal (95% CI)	306	248		47.9 %	0.51 [ 0.14, 1.82 ]
Total events: 21 (Antibiotic), Heterogeneity: Tau <sup>2</sup> = 0.64; 6 Test for overall effect: $Z = 1.6$ 3 Pre- and post-operative pro-	21 (Placebo) Chi <sup>2</sup> = 3.96, df = 1 (P = 04 (P = 0.30) ophylaxis	= 0.05); I <sup>2</sup> =75%			
Bystedt 1981	5/40	7/20		20.5 %	0.36 [ 0.13, 0.98 ]
Subtotal (95% CI)	40	20	•	20.5 %	0.36 [ 0.13, 0.98 ]
Total events: 5 (Antibiotic), 7 Heterogeneity: not applicable Test for overall effect: $Z = 1.5$	(Placebo) e 99 (P = 0.047)				
Total (95% CI)	390	285	•	100.0 %	0.60 [ 0.32, 1.11 ]
Total events: 46 (Antibiotic), Heterogeneity: Tau <sup>2</sup> = 0.23; Test for overall effect: $Z = 1$ , Test for subgroup differences	36 (Placebo) Chi <sup>2</sup> = 7.05, df = 3 (P = 62 (P = 0.10) : Chi <sup>2</sup> = 3.04, df = 2 (P	= 0.07); l <sup>2</sup> =57% = 0.22), l <sup>2</sup> =34%			
			0.01 0.1 10 100		
			Favours antibiotic Favours placebo		

## Analysis I.4. Comparison I Antibiotic versus placebo, Outcome 4 Pain score (VAS 7th day).

Review: Antibiotics to prevent complications following tooth extractions

Comparison: I Antibiotic versus placebo

Outcome: 4 Pain score (VAS 7th day)

Study or subgroup	Mean Difference (SE)	Mean Difference IV,Random,95% Cl	Weight	Mean Difference IV,Random,95% CI
I Pre-operative prophylaxis				
Kaczmarzyk 2007	-0.63 (6.9057)		7.6 %	-0.63 [ -14.16, 12.90 ]
Bezerra 2011	-15.3 (4.6368)		16.8 %	-15.30 [ -24.39, -6.21 ]
L pez-Cedr n 2011	-4.1 (4.5881)		17.2 %	-4.10 [ -13.09, 4.89 ]
Subtotal (95% CI) Heterogeneity: Tau <sup>2</sup> = 32.18 Test for overall effect: $Z = 1$ .	Chi <sup>2</sup> = 4.34, df = 2 (P = 0.11); l <sup>2</sup> =549 56 (P = 0.098)	%	41.6 %	-7.41 [ -16.18, 1.36 ]
Barclay 1987	-8.46 (3.7656)		25.5 %	-8.46 [ -15.84, -1.08 ]
Kaczmarzyk 2007	-6.22 (5.5403)		11.8 %	-6.22 [ -17.08, 4.64 ]
L pez-Cedr n 2011	-9.27 (4.1415)	-	21.1 %	-9.27 [ -17.39, -1.15 ]
Subtotal (95% CI)		•	58.4 %	-8.30 [ -13.18, -3.42 ]
Test for overall effect: $Z = 3$ . <b>Total (95% CI)</b> Heterogeneity: Tau <sup>2</sup> = 0.0; C Test for overall effect: $Z = 4$ . Test for subgroup differences	$hi^{2} = 4.54, df = 5 (P = 0.47); l^{2} = 0.0\%$ $B0 (P = 0.0000 l 7); l^{2} = 0.03, df = 1 (P = 0.86), l^{2} = 0.0000 l 7; l^{2} = 0.0000 l 7; l^{2} = 0.0000 l 7; l^{2} = 0.00000 l 7; l^{2} = 0.000000000; l^{2} = 0.0000000000; l^{2} = 0.000000000; l^{2} = 0.0000000000; l^{2} = 0.0000000000; l^{2} = 0.00000000000; l^{2} = 0.00000000000; l^{2} = 0.00000000000; l^{2} = 0.000000000000; l^{2} = 0.000000000000; l^{2} = 0.000000000000; l^{2} = 0.0000000000000; l^{2} = 0.000000000000; l^{2} = 0.000000000000; l^{2} = 0.0000000000; l^{2} = 0.0000000000000; l^{2} = 0.00000000000000; l^{2} = 0.00000000000000; l^{2} = 0.00000000000000000; l^{2} = 0.0000000000000000; l^{2} = 0.00000000000000; l^{2} = 0.000000000000000; l^{2} = 0.000000000000000; l^{2} = 0.0000000000000000000; l^{2} = 0.0000000000000000000; l^{2} = 0.000000000000000000000000000000000; l^{2} = 0.0000000000000000000000000000000000$	*	100.0 %	-8.17 [ -11.90, -4.45 ]
	Fa	-50 -25 0 25 50 vours antibiotic Favours placebo		

#### Analysis 1.5. Comparison I Antibiotic versus placebo, Outcome 5 Fever (6th-7th day).

Review: Antibiotics to prevent complications following tooth extractions

Comparison: I Antibiotic versus placebo

Outcome: 5 Fever (6th-7th day)

Study or subgroup	Antibiotic	Placebo		F	isk Ratio M-		Weight	Risk Ratio
	n/N	n/N		H,Ran	dom,95% Cl			H,Random,95% Cl
Lacasa 2007 (1)	0/68	0/62						Not estimable
Arteagoitia 2005 (2)	1/259	8/231		-			37.7 %	0.11 [ 0.01, 0.88 ]
Bystedt 1981 (3)	0/40	0/20						Not estimable
Happonen 1990 (4)	8/9	6/45					62.3 %	0.66 [ 0.24, 1.79 ]
Total (95% CI)       458       358         Total events: 9 (Antibiotic), 14 (Placebo)         Heterogeneity: Tau <sup>2</sup> = 1.05; Chi <sup>2</sup> = 2.53, df = 1 (P = 0.11); l <sup>2</sup> =60%         Test for overall effect: Z = 1.20 (P = 0.23)         Test for subgroup differences: Not applicable				-	-		100.0 %	0.34 [ 0.06, 1.99 ]
			0.0 I Favours ar	0.1 ntibiotic	10 Favours	100 placebo		

(I) Preoperative antibiotics

(2) Post-operative antibiotics

(3) Pre- and post-operative antibiotics

(4) Pre- and post-operative antibiotics

## Analysis I.6. Comparison I Antibiotic versus placebo, Outcome 6 Swelling (7th day).

Review: Antibiotics to prevent complications following tooth extractions

Comparison: I Antibiotic versus placebo

Outcome: 6 Swelling (7th day)

Study or subgroup	Antibiotic	Placebo	Risk Ratio M-	Weight	Risk Ratio M-
	n/N	n/N	H,Random,95% Cl		H,Random,95% Cl
l Pre-operative prophylaxis					
Sekhar 2001	21/44	7/17	-	29.2 %	1.16[0.61, 2.21]
Kaczmarzyk 2007	9/31	2/14	<b></b>	6.3 %	2.03 [ 0.50, 8.21 ]
L pez-Cedr n 2011	10/39	6/20	-	16.7 %	0.85 [ 0.36, 2.01 ]
Subtotal (95% CI)	114	51	•	52.2 %	1.13 [ 0.69, 1.83 ]
Total events: 40 (Antibiotic), 12 Heterogeneity: $Tau^2 = 0.0$ ; Chi Test for overall effect: $Z = 0.48$ 2 Post-operative prophylaxis	5 (Placebo) i <sup>2</sup> = 1.11, df = 2 (P = 3 (P = 0.63)	0.57); I <sup>2</sup> =0.0%			
Sekhar 2001	11/47	8/17		23.4 %	0.50 [ 0.24, 1.02 ]
Subtotal (95% CI)	47	17	•	23.4 %	0.50 [ 0.24, 1.02 ]
Iotal events: 11 (Antibiotic), 8 Heterogeneity: not applicable Test for overall effect: Z = 1.90 3 Pre- and post-operative prop Kaczmarzyk 2007	(Placebo) 0 (P = 0.058) ohylaxis 6/28	2/13		57%	1 39 [ 0 32 5 99 ]
L poz Codr. p 2011	12/44	6/20	_	19.4 %	
	70	22		24 4 9/	1.07[0.53, 2.17]
Subtotal (95% CI) Total events: 19 (Antibiotic), 8 Heterogeneity: Tau <sup>2</sup> = 0.0; Chi	/2 (Placebo) $i^2 = 0.17, df = 1$ (P =	<b>33</b> 0.68); I <sup>2</sup> =0.0%		24.4 %	1.0/ [ 0.53, 2.1/ ]
Total (95% CI)	<b>233</b>	101	•	100.0 %	0.92 [ 0.65, 1.30 ]
Total events: 70 (Antibiotic), 3 Heterogeneity: Tau <sup>2</sup> = 0.0; Chi Test for overall effect: $Z = 0.48$ Test for subgroup differences: 0	I (Placebo) $i^2 = 4.95$ , df = 5 (P = 3 (P = 0.63) Chi <sup>2</sup> = 3.62, df = 2 (F	0.42); l <sup>2</sup> =0.0% P = 0.16), l <sup>2</sup> =45%			- · ·
			0.01 0.1 10 100		
			Favours antibiotic Favours placebo		

## Analysis I.7. Comparison I Antibiotic versus placebo, Outcome 7 Trismus (7th day).

Review: Antibiotics to prevent complications following tooth extractions

Comparison: I Antibiotic versus placebo

Outcome: 7 Trismus (7th day)

Study or subgroup	Antibiotic	Placebo	I	Risk Ratio	Weight	Risk Ratio
	n/N	n/N	H,Rar	ndom,95% Cl		H,Random,95% Cl
Kaczmarzyk 2007 (I)	7/31	3/14			34.5 %	1.05 [ 0.32, 3.49 ]
Pasupathy 2011 (2)	5/60	4/29			32.3 %	0.60 [ 0.18, 2.08 ]
Kaczmarzyk 2007 (3)	6/28	3/13	-4	<b>-</b>	33.2 %	0.93 [ 0.27, 3.14 ]
<b>Total (95% CI)</b> Total events: 18 (Antibiotic), Heterogeneity: Tau <sup>2</sup> = 0.0; C Test for overall effect: Z = 0.0 Test for subgroup differences	<b>119</b> 10 (Placebo) hi <sup>2</sup> = 0.44, df = 2 (P = 47 (P = 0.64) : Not applicable	<b>56</b> = 0.80); I <sup>2</sup> =0.0%	-		100.0 %	0.84 [ 0.42, 1.71 ]
			0.01 0.1 Favours antibiotic	10 100 Favours placebo		
(1) Preoperative antibiotics						
(2) Pre-operative antibiotics						

#### Analysis I.8. Comparison I Antibiotic versus placebo, Outcome 8 Adverse effects.

Review: Antibiotics to prevent complications following tooth extractions

Comparison: I Antibiotic versus placebo

Outcome: 8 Adverse effects

Study or subgroup	Antibiotic	Placebo	Risk Ratio	Weight	Risk Ratio
	n/N	n/N	H,Random,95% Cl		H,Random,95% Cl
Kaczmarzyk 2007 (I)	0/31	0/14			Not estimable
Lacasa 2007 (2)	1/68	1/31		4.5 %	0.46 [ 0.03, 7.05 ]
Arteagoitia 2005 (3)	14/259	2/231		14.5 %	6.24 [ 1.43, 27.18 ]
Lacasa 2007 (4)	10/69	2/31		14.7 %	2.25 [ 0.52, 9.65 ]
Bystedt 1981 (5)	0/40	0/20			Not estimable
Barclay 1987 (6)	20/45	14/50	<b>-</b>	62.3 %	1.59 [ 0.91, 2.76 ]
Kaczmarzyk 2007 (7)	3/28	0/13		4.1 %	3.38 [ 0.19, 61.03 ]
Total (95% CI)	540	390	◆	100.0 %	1.98 [ 1.10, 3.59 ]
Total events: 48 (Antibiotic),	19 (Placebo)				
Heterogeneity: Tau <sup>2</sup> = 0.07; 0	Chi <sup>2</sup> = 4.50, df = 4 (P	= 0.34);   <sup>2</sup> =   %			
Test for overall effect: $Z = 2.2$	27 (P = 0.023)				
Test for subgroup differences:	Not applicable				

0.01 0.1 10 100 Favours placebo

Favours antibiotic

(1) Pre-operative antibiotics

(2) Pre-operative antibiotics

(3) Post-operative antibiotics

(4) Post-operative antibiotics

(5) Pre- and post-operative antibiotics

(6) Pre- and post-operative antibiotics

(7) Pre- and post-operative antibiotics

## ADDITIONAL TABLES

Author	Country	% for caries	% for periodontitis
Da'ameh 2006	Afghanistan	59.2	35.3
Akhter 2008	Bangladesh	67.5	18.5
Jovino-Silveira 2005	Brazil	63.3	13.1
Chrysanthakopoulos 2011	Greece	45.6	32.1
Anand 2010	India	44.6	33.2
Aida 2009	Japan	43.6	37.1
Baqain 2007	Jordan	63.8	22.9
Al-Shammari 2006	Kuwait	43.7	37.4
Byahatti 2011	Libya	55.9	34.4
Danielson 2011	Nigeria	32.6	45
Trovik 2000	Norway	40	24
Chestnutt 2000	Scotland	51	21
McCaul 2001	Scotland	54.7	16.7
Lesolang 2009	South Africa	47.9	22.6
Richards 2005	Wales	59	29.1

#### Table 1. Studies of reasons for tooth extraction

Table 2. Kaziro 1984 outcome data (from graphs)

	Antibiotic group	Placebo group
Infection- wound breakdown	5%	27%
Pain - Day 8	60% pain free	22% pain free
Swelling - Day 8	80% minimal swelling	66% minimal swelling

#### Table 3. Raw outcome data - Local signs of infection

Infection (%)				
	Antibiotic	Placebo		
Pre-operative prophylaxis				
Mitchell 1986	4/25 (16%)	14/25 (56%)		
Sekhar 2001	1/44 (2%)	0/17		
Lacasa 2007	4/75 (5%)	6/38 (16%)		
Halpern 2007	0/59	5/59 (8%)		
Bezerra 2011	0/34	0/34		
López-Cedrún 2011	0/39	3/20 (15%)		
Pasupathy 2011	2/60 (3%)	3/29 (10%)		
Post-operative prophylaxis				
Sekhar 2001	0/47	0/17		
Arteagoitia 2005	5/259 (2%)	30/231 (13%)		
Lacasa 2007	2/72 (3%)	6/37 (16%)		
López-Cedrún 2011	0/44	2/20 (10%)		
Pre- and post-operative prophylaxis				
Happonen 1990	11/91 (12%)	5/45 (11%)		
Leon Arcila 2001	0/49	0/53		
Mean incidence of infection per group	29/898 (3.4%)	74/625 (12.5%)		

## Table 4. Raw outcome data - Dry socket

Antibiotic	Placebo
A	Antibiotic

#### Table 4. Raw outcome data - Dry socket (Continued)

Ritzau 1992	6/135 (4.4%)	7/135 (5.2%)
Bergdahl 2004*	10/59 (16.9%)	13/60 (21.6%)
Kaczmarzyk 2007	1/31 (3.2%)	2/13 (15.4%)
Halpern 2007	0/59	0/59
Bezerra 2011	0/34	0/34
López-Cedrún 2011	0/39	0/20
Post-operative prophylaxis		
Arteagoitia 2005	0/259	2/231 (0.9%)
López-Cedrún 2011	0/44	0/20
Pre- and post-operative prophylaxis		
Bystedt 1981	2/40 (5%)	2/20 (1%)
Barclay 1987*	8/45 (17.8%)	17/50 (34%)
Kaczmarzyk 2007	2/28 (7.1%)	2/14 (14.3%)
Mean incidence dry socket per group	29/773 (3.8%)	45/656 (6.9%)

\* Participants in both these studies had some pericoronitis in the recent past and were therefore at higher risk of infection.

## Table 5. Raw outcome data - VAS pain scores

Mean (SD) VAS pain scores				
	Antibiotic	Placebo		
Pre-operative prophylaxis				
Kaczmarzyk 2007	9.51 (28.83) n = 31	10.14 (19.54) n = 13		
López-Cedrún 2011	10.92 (16.89) n = 39	5.75 (12.26) n = 20		
Bezerra 2011	15.9 (23.6) n = 34	31.2 (29.5) n = 34		

Antibiotics to prevent complications following tooth extractions (Review)

Copyright © 2013 The Cochrane Collaboration. Published by John Wiley & Sons, Ltd.

#### Table 5. Raw outcome data - VAS pain scores (Continued)

Pre- and post-operative prophylaxis		
Barclay 1987*	6.33 (12.58) n = 45	14.79 (23.09) n = 50
Kaczmarzyk 2007	3.92 (9.79) n = 28	10.14 (19.54) n = 14
López-Cedrún 2011	15.02 (23.44) n = 44	5.75 (12.26) n = 20

\*Participants in this study had some pericoronitis in the recent past and were therefore at higher risk of infection. SD = standard deviation; VAS = visual analogue scale.

#### APPENDICES

#### Appendix I. Cochrane Oral Health Group's Trials Register search strategy

((extract\* or remov\* or exodontia or "impacted teeth" or "impacted tooth" or "oral surg\*" or (tooth and surg\*) or (teeth and surg\*) or ("third molar\*" and surg\*)) AND (antibiotic\* or erthromycin\* or metronidaz\* or tetracycline\* or clindamycin\* or teicoplanin\* or vancomycin\* or floxacillin\* or gentamicin\* or cephalexin\* or "anti biotic\*" or anti-biotic\* or penicillin\* or antibacterial\* or antibacterial\* or antibacterial\*" or erthromycin\* or cephalesporin\* or suphonamide\* or clindamicin\* or augmentin\* or floxacillin\* or anti-fungal\* or "anti fungal\*" or vancomicin\* or flucloxacillin\* or floxacillin\* or gentamycin\* or cephalexin\*))

#### Appendix 2. CENTRAL search strategy

- #1 MeSH descriptor Tooth extraction explode all trees
- #2 exodontia in All Text

#3 ((tooth in All Text near/4 extract\* in All Text) or (teeth in All Text near/4 extract\* in All Text) or ("third molar\*" in All Text near/4 extract\* in All Text) or (3rd in All Text and (molar\* in All Text near/4 extract\* in All Text) ) or "dental extract\*" in All Text or (tooth in All Text near/4 remov\* in All Text) or (teeth in All Text near/4 remov\* in All Text) or ("third molar\*" in All Text near/4 remov\* in All Text) or ("3rd molar\*" in All Text near/4 remov\* in All Text) or (tooth in All Text near/4 surg\* in All Text) or (teeth in All Text near/4 remov\* in All Text) or ("3rd molar\*" in All Text near/4 remov\* in All Text) or (tooth in All Text near/4 surg\* in All Text) or (teeth in All Text near/4 surg\* in All Text) or ("third molar\*" in All Text near/4 surg\* in All Text) or ("3rd molar\*" in All Text near/4 surg\* in All Text))

- #4 (#1 or #2 or #3)
- #5 MeSH descriptor Molar explode all trees
- #6 MeSH descriptor Tooth, impacted this term only
- #7 ("wisdom tooth" in All Text or "wisdom teeth" in All Text or (third in All Text near/3 molar in All Text))
- #8 "impacted tooth" in All Text
- #9 "impacted teeth" in All Text
- #10 (#5 or #6 or #7 or #8 or #9)
- #11 (extract\* in All Text or remov\* in All Text or surg\* in All Text)
- #12 (#10 and #11)
- #13 (#4 or #12)

- #14 MeSH descriptor Anti-Bacterial Agents explode all trees
- #15 MeSH descriptor Antibiotic prophylaxis this term only
- #16 MeSH descriptor Erythromycin explode all trees
- #17 MeSH descriptor Metronidazole this term only
- #18 MeSH descriptor Tetracyclines this term only
- #19 MeSH descriptor Clindamycin this term only
- #20 MeSH descriptor Teicoplanin this term only
- #21 MeSH descriptor Vancomycin this term only
- #22 MeSH descriptor Floxacillin this term only
- #23 MeSH descriptor Gentamicins this term only
- #24 MeSH descriptor Cephalexin this term only
- #25 (antibiot\* in All Text or "anti biot\*" in All Text or anti-biot\* in All Text)
- #26 (penicillin\* in All Text or erythromycin\* in All Text or metronidazol\* in All Text or cephalosporin\* in All Text)

#27 (sulphonamide\* in All Text or tetracycline\* in All Text or clindamycin\* in All Text or clindamicin\* in All Text or augmentin\* in All Text or flagyl\* in All Text or amoxyl\* in All Text or amoxil\* in All Text or co-amox\* in All Text or antifungal\* in All Text or antifungal\* in All Text or teicoplanin\* in All Text or vancomycin\* in All Text or vancomicin\* in All Text or fluctoracilin\* in All Text or flow or fl

#28 (antibacterial in All Text or anti-bacterial in All Text or "anti bacterial" in All Text)

#29 (#14 or #15 or #16 or #17 or #18 or #19 or #20 or #21 or #22 or #23 or #24 or #25 or #26 or #27 or #28)

#30 (#13 and #29)

## Appendix 3. MEDLINE via OVID search strategy

- 1. exp TOOTH EXTRACTION/
- 2. exodontia.mp.

3. ((tooth adj4 extract\$) or (teeth adj4 extract\$) or ("third molar\$" adj4 extract\$) or ("3rd molar\$" adj4 extract\$) or "dental extract\$" or (tooth adj4 remov\$) or (teeth adj4 remov\$) or ("third molar\$" adj4 remov\$) or ("third molar\$" adj4 remov\$) or (tooth adj4 surg\$) or (teeth adj4 surg\$) or ("third molar\$" adj4 surg\$) or ("3rd molar\$" adj4 surg\$)).mp.

- 4. 1 or 2 or 3
- 5. MOLAR/
- 6. TOOTH IMPACTED/
- 7. ("wisdom tooth" or "wisdom teeth" or (third adj3 molar\$) or (3rd adj3 molar\$)).mp.
- 8. "impacted tooth".mp.
- 9. "impacted teeth".mp.
- 10. (5 or 6 or 7 or 8 or 9) and (extract\$ or remov\$ or surg\$).mp.
- 11. 10 or 4
- 12. exp ANTIBIOTICS/
- 13. ANTIBIOTIC PROPHYLAXIS/
- 14. ERYTHROMYCIN/
- 15. METRONIDAZOLE/
- 16. TETRACYCLINES/
- 17. CLINDAMYCIN/
- 18. TEICOPLANIN/
- 19. VANCOMYCIN/
- 20. FLOXACILLIN/
- 21. GENTAMICINS/
- 22. CEPHALEXIN/
- 23. (Antibiot\$ or "anti biot\$" or anti-biot\$).mp.
- 24. (penicillin\$ or erythromycin\$ or Metronidazol\$ or Cephalosporin\$).mp.
- 25. (antibacterial or anti-bacterial or "anti bacterial").mp.

26. (sulphonamide\$ or tetracycline\$ or clindamycin\$ or clindamicin\$ or augmentin\$ or flagyl\$ or amoxyl\$ or amoxil\$ or coamox\$ or antifungal\$ or anti-fungal\$ or "anti fungal\$" or teicoplanin\$ or vancomycin\$ or vancomicin\$ or flucloxacillin or floxacillin or gentamycin\$ or gentamycin\$ or cephalexin\$).mp.

Antibiotics to prevent complications following tooth extractions (Review)

Copyright © 2013 The Cochrane Collaboration. Published by John Wiley & Sons, Ltd.

#### 27. or/12-26

28. 27 and 11

The above subject search was linked to the Cochrane Highly Sensitive Search Strategy (CHSSS) for identifying randomised trials in MEDLINE: sensitivity maximising version (2008 revision) as referenced in Chapter 6.4.11.1 and detailed in box 6.4.c of the *Cochrane Handbook for Systematic Reviews of Interventions* version 5.1.0 (updated March 2011).

1. randomized controlled trial.pt.

- 2. controlled clinical trial.pt.
- 3. randomized.ab.
- 4. placebo.ab.
- 5. drug therapy.fs.
- 6. randomly.ab.
- 7. trial.ab.
- 8. groups.ab.
- 9. or/1-8
- 10. exp animals/ not humans.sh.
- 11. 9 not 10

#### Appendix 4. EMBASE via OVID search strategy

- 1. exp TOOTH EXTRACTION/
- 2. exodontia.ti,ab.

3. ((tooth adj4 extract\$) or (teeth adj4 extract\$) or ("third molar\$" adj4 extract\$) or ("3rd molar\$" adj4 extract\$) or "dental extract\$" or (tooth adj4 remov\$) or (teeth adj4 remov\$) or ("third molar\$" adj4 remov\$) or ("third molar\$" adj4 remov\$) or (tooth adj4 surg\$) or (teeth adj4 surg\$) or ("third molar\$" adj4 surg\$) or ("ard molar\$" adj4 surg\$)).ti,ab.

- 4. 1 or 2 or 3
- 5. exp TOOTH/
- 6. ("wisdom tooth" or "wisdom teeth" or (third adj3 molar\$) or (3rd adj3 molar\$)).ti,ab.
- 7. "impacted tooth".ti,ab.
- 8. "impacted teeth".ti,ab.
- 9. or/5-8
- 10. (extract\$ or remov\$ or surg\$).ti,ab.
- 11. 9 and 10
- 12. 4 or 11
- 13. exp ANTIBIOTIC AGENT/
- 14. ANTIBIOTIC PROPHYLAXIS/
- 15. METRONIDAZOLE/
- 16. (Antibiot\$ or "anti biot\$" or anti-biot\$).ti,ab.
- 17. (penicillin\$ or erythromycin\$ or Metronidazol\$ or Cephalosporin\$).ti,ab.
- 18. (antibacterial or anti-bacterial or "anti bacterial").ti,ab.
- 19. (sulphonamide\$ or tetracycline\$ or clindamycin\$ or clindamicin\$ or anti-fungal\$ or "anti fungal\$" or teicoplanin\$ or vancomycin\$ or vancomicin\$ or flucloxacillin or floxacillin or gentamycin\$ or gentamycin\$ or cephalexin\$).ti,ab.
- 20. or/13-19
- 21. 12 and 20

The above subject search was linked to the Cochrane Oral Health Group filter for EMBASE via OVID:

- 1. random\$.ti,ab.
- 2. factorial\$.ti,ab.
- 3. (crossover\$ or cross over\$ or cross-over\$).ti,ab.
- 4. placebo\$.ti,ab.
- 5. (doubl\$ adj blind\$).ti,ab.
- 6. (singl\$ adj blind\$).ti,ab.
- 7. assign\$.ti,ab.
- 8. allocat\$.ti,ab.
- 9. volunteer\$.ti,ab.

10. CROSSOVER PROCEDURE.sh.
 11. DOUBLE-BLIND PROCEDURE.sh.
 12. RANDOMIZED CONTROLLED TRIAL.sh.
 13. SINGLE BLIND PROCEDURE.sh.
 14. or/1-13
 15. ANIMAL/ or NONHUMAN/ or ANIMAL EXPERIMENT/
 16. HUMAN/
 17. 16 and 15
 18. 15 not 17
 19. 14 not 18

#### Appendix 5. LILACS via BIREME search strategy

(Mh Tooth extraction or Mh Extracción Dental or Mh Extração Dentária or ((Tw tooth or Tw teeth or Tw molar\$ or Tw dental) and (Tw extrac\$ or Tw remov\$ or Tw surg\$))) [Words] and (Mh Anti-Bacterial Agents or Mh Agentes Antibacterianos or Mh Antibiotic Prophylaxis or Profilaxis Antibiótica or Mh Antibioticoprofilaxia or antibiot\$ or "anti biot\$" or anti-biot\$ or anti-bacte\$ or anti-bacte\$ or "anti bacte\$" or penicillin\$ or erythromycin\$ or metronidazol\$ or cephalosporin\$ or sulphonamide\$ or tetracycline\$ or clindamycin\$ or clindamicin\$ or augmentin\$ or flagyl\$ or amoxyl\$ or amoxil\$ or co-amox\$ or teicoplanin\$ or vancomycin\$ or vancomicin\$ or flucloxacillin\$ or floxacillin\$ or gentamicin\$ or gentamycin\$ or cephalexin\$) [Words]

The above subject search was linked to the Brazilian Cochrane Center filter for LILACs via BIREME:

Pt randomized controlled trial OR Pt controlled clinical trial OR Mh randomized controlled trials OR Mh random allocation OR Mh double-blind method OR Mh single-blind method) AND NOT (Ct animal AND NOT (Ct human and Ct animal)) OR (Pt clinical trial OR Ex E05.318.760.535\$ OR (Tw clin\$\$ AND (Tw trial\$\$ OR Tw ensa\$\$ OR Tw estud\$\$ OR Tw experim\$\$ OR Tw investiga\$)) OR ((Tw singl\$\$ OR Tw simple\$\$ OR Tw doubl\$\$ OR Tw doubl\$\$ OR Tw duplo\$\$ OR Tw trebl\$\$ OR Tw trip\$\$) AND (Tw blind\$\$ OR Tw cego\$\$ OR Tw ciego\$\$ OR Tw mask\$\$ OR Tw masca\$\$)) OR Mh placebos OR Tw placebos OR (Tw random\$\$ OR Tw randon\$\$ OR Tw casual\$\$ OR Tw acaso\$\$ OR Tw azar OR Tw aleator\$\$) OR Mh research design) AND NOT (Ct animal AND NOT (Ct human and Ct animal)) OR (Ct comparative study OR Ex E05.337\$\$ OR Mh follow-up studies OR Mh prospective studies OR Tw control\$\$ OR Tw prospectiv\$\$ OR Tw volunt\$\$ OR Tw volunteer\$\$ AND NOT (Ct animal AND NOT (Ct human and Ct animal))] and not (Ct ANIMAL AND NOT (Ct HUMAN and Ct ANIMAL))]

## FEEDBACK

#### Comment from Dr John Curran, 11 February 2013

#### Summary

Very good review. Unfortunately the difficulty of designing a randomised trial still exists partly due to ethical requirements and logistics -e.g. with third molar surgery assessment of difficulty and surgical ability are hard to measure. Post-operative assessment also needs to be done sooner than the 7 days used in the review. Little has changed in but I believe that in patient age groups most prevalent in the North America context the incidence of infection is even lower than reported -i.e. antibiotic usage should be highly selective.

#### Reply

Thank you for your interest in our work and for your comment.

I absolutely agree that real incidence of infectious complications in a population similar to study groups is likely to be lower, and for that reason we did not really recommend for regular antibiotic prophylaxis. Unfortunately because of the lack of studies on patients at higher risk, no evidence is available on cases for which antibiotic prophylaxis is (anecdotally) recommended.

#### Contributors

Summary: John Curran. Reply: Giovanni Lodi.

#### WHAT'S NEW

Last assessed as up-to-date: 25 January 2012.

Date	Event	Description
24 April 2013	Feedback has been incorporated	Comment from Dr John Curran.

#### CONTRIBUTIONS OF AUTHORS

- Conceiving, designing and co-ordinating the review: Giovanni Lodi (GL).
- Screening search results and retrieved papers against inclusion criteria: GL, Susan Furness (SF), Lara Figini (LF).
- Appraising quality of papers: GL, SF.
- Extracting data from papers: GL SF, Andrea Sardella (AS).
- Writing to authors of papers for additional information: GL, SF.
- Data management for the review and entering data into RevMan: GL, SF, LF.
- Analysis and interpretation of data: GL, SF, AS.
- Preparing the summary of findings table: SF.
- Providing a clinical perspective: GL, Antonio Carrassi (AC), Massimo Del Fabbro (MD).
- Writing the text of the review: GL, SF.
- Preparing the plain language summary: GL.
- Providing comments on a draft of the review: AS, MD.

## DECLARATIONS OF INTEREST

None known.

## SOURCES OF SUPPORT

#### **Internal sources**

• MAHSC, UK.

The Cochrane Oral Health Group is supported by the Manchester Academic Health Sciences Centre (MAHSC) and the NIHR Manchester Biomedical Research Centre.

• The University of Manchester, UK.
### **External sources**

• Cochrane Oral Health Group Global Alliance, UK.

All reviews in the Cochrane Oral Health Group are supported by Global Alliance member organisations (British Orthodontic Society, UK; British Society of Paediatric Dentistry, UK; National Center for Dental Hygiene Research & Practice, USA and New York University College of Dentistry, USA) providing funding for the editorial process (http://ohg.cochrane.org/).

• National Institute for Health Research (NIHR), UK.

CRG funding acknowledgement:

The NIHR is the largest single funder of the Cochrane Oral Health Group.

Disclaimer:

The views and opinions expressed therein are those of the authors and do not necessarily reflect those of the NIHR, NHS or the Department of Health.

# DIFFERENCES BETWEEN PROTOCOL AND REVIEW

Quasi-randomised studies are no longer eligible for inclusion in this review because less biased evidence is available from randomised controlled trials.

It was decided to include only double-blind placebo-controlled studies because we believed that these studies were likely to provide the best evidence to inform practice.

We clarified excluded outcomes by specifically excluding trials where the outcome was endocarditis incidence, bacteraemia or serum marker of infection only.

## INDEX TERMS

### Medical Subject Headings (MeSH)

\*Antibiotic Prophylaxis [adverse effects]; Anti-Bacterial Agents [adverse effects; \*therapeutic use]; Controlled Clinical Trials as Topic; Dry Socket [prevention & control]; Molar, Third [\*surgery]; Pain, Postoperative [prevention & control]; Tooth Extraction [\*adverse effects]; Tooth, Impacted [\*surgery]

### MeSH check words

Humans